JRC Scientific and Technical Reports



ICT and Regional Economic Dynamics: A Literature Review

Authors: Charlie Karlsson, Gunther Maier, Michaela Trippl, Iulia Siedschlag, Robert Owen and Gavin Murphy

Editors: Andrea de Panizza and Marc Bogdanowicz



EUR 24510 EN - 2010





The mission of the JRC-IPTS is to provide customer-driven support to the EU policy-making process by developing science-based responses to policy challenges that have both a socio-economic as well as a scientific/technological dimension.

European Commission Joint Research Centre Institute for Prospective Technological Studies

Contact information

Address: Edificio Expo. c/ Inca Garcilaso, 3. E-41092 Seville (Spain)

E-mail: jrc-ipts-secretariat@ec.europa.eu

Tel.: +34 954488318 Fax: +34 954488300

http://ipts.jrc.ec.europa.eu http://www.jrc.ec.europa.eu

Legal Notice

Neither the European Commission nor any person acting on behalf of the Commission is responsible for the use which might be made of this publication.

Europe Direct is a service to help you find answers to your questions about the European Union

Freephone number (*): 00 800 6 7 8 9 10 11

(*) Certain mobile telephone operators do not allow access to 00 800 numbers or these calls may be billed.

A great deal of additional information on the European Union is available on the Internet.

It can be accessed through the Europa server http://europa.eu/

JRC 59920

EUR 24510 EN ISBN 978-92-79-16568-9 ISSN 1018-5593 doi:10.2791/46419

Luxembourg: Publications Office of the European Union

© European Union, 2010

Reproduction is authorised provided the source is acknowledged

Printed in Spain

Preface

This paper was prepared as part of the study "The knowledge economy, economic transformations and ICT in the EU25+: Regional dynamics in the deployment phase". The Institute for Prospective Technological Studies of the European Commission's Joint Research Centre commissioned the IGICT consortium to carry out the study and write the report. This consortium was coordinated by Professor Iulia Siedschlag of the Economic and Social Research Institute (ESRI – Ireland) and Professor Dr. Gunther Maier, of the Vienna University of Economic and Business Administration (VUW – Austria).

The consortium was made up of the following members:

Coordination

Professor Iulia Siedschlag, Economic and Social Research Institute (ESRI – Ireland.

Professor Dr. Gunther Maier, Vienna University of Economic and Business Administration (VUW – Austria).

Partners

Professor Robert Owen, University of Nantes, France.

Professor Andres Rodrigues-Pose, London School of Economics and Political Science, UK.

Professor Charlie Karlsson, Jönköping University, Sweden.

Professor Gerhrad Untiedt, Director, GEFRA, Germany.

Professor Markku Wilenius, Director, Turku School of Economics and Business Administration, Finland.

Professor Roberta Capello, Polietcnico of Milan, Italy.

Professor (Associate) Grogori Fainstein, Tallin University of Technology, Estonia.

Professor Janusz Zaleski, President, Wroclaw Regional Development Agency, Poland.

Professor (Associate) Joze Damijan, University of Ljubljana, Slovenia.

The report has also been published as an ESRI Working paper (No. 233).

Table of Contents

Pre	tace	1
Exe	cutive Summary	5
1.	Introduction	9
2	ICT: Definitions and Measures	. 11
3.2	The Economic Impact of ICT Diffusion	. 17 . 19
4.2	Regional Dynamics and Economic Transformations during the Deployment Phase of ICT	. 29 . 33
5 5.1 5.2	Regional Innovation Systems and the Layers of Innovation	43
6.2	Globalisation and ICT in the Knowledge-based Economy	63 66
7 7.1 7.2	Summary and Conclusions	69
Ref	erences	81

Executive Summary

Information and Communication Technologies (ICTs) undoubtedly constitute one of the *key innovations* of the last century. ICT represent a new technological paradigm that belongs to the family of General Purpose Technologies (GPTs). A GPT has the potential to be pervasively adopted and adapted to a many sectors of the economy in ways that drastically change operations and products and the relationships between different sectors. The characteristics of GPTs have been described by Bresnahan & Trajtenberg (1995, p.84): "*Most GPTs play the role of 'enabling technologies', opening up new opportunities rather than offering complete, final solutions.*" GPTs also involve 'innovational complementarities', i.e. "the productivity of R&D in a downstream sector increases as a consequence of innovation in the GPT technology". Thus, GPTs have two major characteristics: *generality of application*; and, *innovational complementarities*. However, other characteristics of GPTs are also important (Lipsey, Becar & Carlaw, 1998): (i) wide scope for improvement initially, (ii) many varied uses, (iii) applicability across large parts of the economy, and (iv) strong complementarities with other technologies.

Some innovations are incremental and some are drastic. ICTs are an example of a drastic innovation, which qualifies as a general purpose technology (GPT), since they have the potential for (i) pervasive use in a wide range of sectors in ways that radically change their modes of operation and the character of their output, (ii) setting the stage for series of incremental innovations, and (iii) producing discontinuities in the observed pattern of resource allocation and the evolution of output. The fact that ICT is a GPT has many implications: i) its adoption entails experimentation, which may lead to innovation by adopting firms, which in turn shows up as total factor productivity growth, ii) as well as innovating themselves, firms can learn from the (successful or unsuccessful) innovation efforts of others, so there are spillover effects (Bresnahan & Trajtenberg, 1995), and iii) successful implementation of an ICT project requires reorganisation of the firm around the new technology (Helpman & Trajtenberg, 1998, Yang & Brynjolffson, 2001; Brynjolfsson, Hitt & Yang, 2002).

ICTs are composed of a wide range of product and service technologies including computer hardware, software and services and a host of telecommunications functions that include wire or wireline, and wireless, satellite products and services. The rapid diffusion of ICT has produced important changes in how and where goods and services are produced, the nature of goods and services produced, and the means by which goods and services are brought to the market and distributed to consumers. This implies that ICT has had an impact on the industrial structure of regions and on the geographical location of different industries not only within the EU but worldwide. ICT has also influenced the relationship between customers and suppliers and the way many markets for intermediate and final goods and services are organised.

However, there are substantial differences among countries and regions, also among the developed economies, as regards their role in the development of ICT and their ability and propensity to adopt ICT applications in various activities and sectors (Johansson, Karlsson & Stough, 2006). This implies, among other things, that there is a substantial variation in the impact of the use of ICT on efficiency, productivity, and economic growth in different countries and regions.

The objective of this paper is to review the relevant theoretical and empirical literature to provide a theoretical and methodological background for the analysis of the consequences of ICT use and globalisation on the regional economies in the European Union (EU).

This review focuses on the following core issues and analytical questions:

ICT: Definitions and Measures

- 1. What do we understand with ICT as a set of technologies and as a sector of the economy?
- 2. How relevant is the existing statistical data for analysing different aspects of the impact and effects of ICT?

The Economic Impact of ICT Diffusion and Regional Dynamics

- 3. What do we know about the effects of ICT diffusion on structural change, productivity and output growth?
- 4. What are the effects of region-specific mechanisms, such as technological and knowledge spillovers, interactions and networking between firms, on ICT diffusion and its impact on the economic efficiency across EU regions?

5. How does ICT influence spatial patterns of economic activity and what type of regional transformations can be linked to ICT?

Regional Innovation Systems and ICT Diffusion

- 6. What key features characterise regional innovation systems in relation to ICT production and use?
- 7. What are the linkages between regional innovation systems and larger institutional frameworks at national and international levels?
- 8. How do regional innovation systems influence the effective use of ICT at regional level?
- 9. How does ICT influence the functioning of regional innovation systems?

Consequences of Globalisation and ICT Diffusion

- 10. What are the consequences of the interactions between ICT use and globalisation on location of economic activities?
- 11. What are the consequences of outsourcing /offshoring driven by globalisation and ICT use on the home and host economies?
- 12. What is the impact of ICT use and globalisation on regional economic performance?

.

1. Introduction

The Europe 2020 strategy puts a renewed emphasis on the potential role ICT can play in order to exit the crisis and prepare the EU economy for the challenges of the next decade. During the last decade, the relation technology/economy has given rise to a wide debate about the emergence of a digital economy, calling for new methods and tools for measurement, while underlining the importance of establishing the right framework conditions.

Such policy framework needs, in turn, to be based on sound analysis of the mechanics of ongoing economic transformations within the context of the globalization process. It has been widely documented that innovation activity is even more spatially concentrated than industrial activity. Also, many have claimed that knowledge spillovers are sharply reduced with distance. Regions and, more specifically, regional innovation systems, are therefore increasingly seen as the natural places to observe the ongoing transformations and structural/technological changes being enabled by ICT. In this context, the European Commission's Communication of March 2009 on ICT R&D and Innovation, which calls for the emergence of ICT poles of excellence, addresses such views from a political angle.

However, little is known about the conditions and consequences of the advent of ICT R&D, production and use on regional economies. Theoretical analysis is needed to better understand the (region-specific) mechanisms at hand.

This report aims to improve current understanding of the nature and dynamics of the economic transformations that are expected to affect European regional economies during the coming decades, and in particular how they relate to two major trends: the deployment of ICT and the globalisation of the economy. It offers a literature review of the impact and determinants of ICT adoption at a regional level. The questions analysed build on the new growth and economic geography theories, and on regional innovation systems. The report explores the following issues:

a- The influence of the advent of ICT on the spatial pattern of economic activities, when considering different types of activities likely to be affected (old vs new, etc.), regional specialization patterns, regional innovation systems, the role played by globalization trends (e.g. international trade, FDI, off-shoring/outsourcing, international knowledge spillovers), etc.

- **b-** The changing nature of different types of externalities (either positive or negative), including learning processes and technological spillovers, backward-forward linkages and labour mobility, knowledge externalities and the transmission costs of both tacit and codified knowledge, etc.
- **c-** The different factors likely to facilitate or hamper the adoption of new technologies such as institutional/policy setting, business environment and dynamism, culture of innovation, access to capital, research activity and infrastructure, labour mobility and education of the workforce, etc.

Structure of the report

The paper is organised as follows. Section 2 provides an overview of definitions and measures with the objective to provide a general conceptual framework for understanding ICT. Section 3 discusses the impact of ICT diffusion on economic performance, in particular the impact of ICT on structural change, productivity and output growth. Section 4 examines the spatial impact of ICT diffusion including ICT-related spatial transformations, the location of economic activities and the impact of ICT on regional economic growth. Section 5 provides an analysis of the key features of regional innovation systems in relation to ICT production and usage. Section 6 discusses the interactions between ICT and globalisation and their consequences on firms' organisational structures, location of economic activities, and economic performance. Section 7 summarises the main theoretical arguments and concepts and articulates an integrated framework for the analysis of the interplay between, ICT diffusion, regional innovation systems, globalisation on regional economies.

2 ICT: Definitions and Measures

At face value, ICT are a collection of technologies and applications, which enable electronic processing, storing, retrieval, and transfer of data to a wide variety of users or clients. According to Cohen, Salomon & Nijkamp (2002), ICT are currently characterised by:

- very dynamic technological changes, with rapid penetration and adoption rates;
- decreasing costs for new equipment and features;
- a rapidly increasing range of applications and penetration in an increasing number of realms of professional and personal life;
- an intertwined institutional market place, with the private sector acting in a decreasingly regulated environment (in most countries);
- a production and services package dependent on a range of qualities of skilled human resources, and
- a convergence of technologies.

Unfortunately, there exists no clear or unambiguous agreement on the definition of ICT, or what sectors should be termed ICT sectors, which has been generally accepted (Schwartz, 1990; Malecki, 1991; Graham & Marvin, 1996). Furthermore, as ICT over time penetrate more and more sectors, more and more sectors deserve to be classified as ICT sectors. However, it is possible to provisionally delineate the providers of ICT in terms i) manufacturing of ICT, ii) wholesale and retail trade of ICT, iii) ICT network services, iv) other ICT services. These providers provide ICT to ICT users, i.e. to households, firms, and public sector organisations.

To analyse the growth effects of ICT it is necessary to measure the extent of *investment in ICT* in the manufacturing sector, in the private service sector and in the public sector. Usually, statistical bureaus estimate investments from surveys among businesses specifically designed to capture investments. Based upon data over investments in well-defined asset groups, such as plant and machinery, infrastructure, dwellings, vehicles, and intangibles, such as R&D, with internationally agreed definitions, is it possible to estimate total investments by

aggregation. However, this doesn't work in the case of investments in ICT due to a lack of generally agreed definitions.¹

To develop such generally agreed definitions for investments in ICT is by no means simple for several reasons. First, the investments are of at least four kinds: hardware, software, network infrastructure (communications equipment), and, in principle also training of personnel² to handle the equipment but only the three first types are normally capitalised. However, even at this relatively aggregated level comparability problems remain (Ahmed, Schreyer & Wölfl, 2004). When software is sold together with hardware, its value may be recorded as either software or included in the hardware value depending upon the circumstances. ICT components included in other products, such as machinery, will not be directly recorded as an investment in ICT.3 The implication is that comparisons of investments in ICT in the manufacturing and the service sector may be biased, since substantial expenditures on ICT in the manufacturing sector might be recorded as intermediate consumption, while they are capitalised as investments in ICT in the service sector. Furthermore, there are large differences between countries regarding the extent to which expenditures for software are capitalised. Implementing ICT in an organisation entails reorganisation costs. These adjustment costs create a stock that yields future benefits. However, this investment is not measured as such in the national accounts (Oulton & Srinivasan, 2005).⁴

Second, we have the rapid improvements in capacities and speed and at the same time a drastic drop in costs of in particular the hardware. In many cases, it isn't enough to measure expenditures on ICT at current prices. Instead a volume measure is needed that controls for changes in the price level of ICT products. Thus, price indices are needed to deflate expenditures at current price to get a 'constant price' measure. Due to rapidly decreasing production costs and strong competition, the prices of key ICT products has fallen drastically in recent decades. At the same time, their capacity has increased rapidly. Obviously, the

The problems of measuring ICT investments and to make comparisons between countries are discussed in Ahmed, Schreyer & Wölfl (2004).

The fast development of ICT implies that the labour force must learn, relearn, train, and retrain, i.e. there is a continuous need for households and for entire economies to make specific investments in human capital as long as standardised ICT solutions are not established in each area.

Focusing exclusively on ICT investment products does not fully reflect the benefits of ICT diffusion within investment products or in the economy at large (Papaconstantinou, Sakurai & Wyckoff, 1996).

The EU KLEMS project provides data on ICT capital assets comparable across a number of OECD countries (see www.euklems.net).

construction of price indices for ICT products is no easy task and a possible source of unreliable results in empirical studies.⁵

There are also other measurement problems involved (Howitt, 1998): (i) many of the knowledge-creating activities are not registered as part of GDP under conventional national accounting, (ii) the contribution of new or improved products to output is typically underestimated, and (iii) the arrival of new technologies makes machines and equipment that were designed for the old one obsolete, and it reduces the value of the skills that workers acquired for the old technology.

However, a time series of investments in ICT in total and in the three major types is not enough to measure the effects of investments in ICT. What is needed is an aggregated, discounted measure of all historic investments in ICT equipment and systems making up the current ICT capital stock,⁶ or rather of the flow of capital services from the stock of ICT capital. However, normally a strict proportionality between capital services and capital stocks but the ratio can vary between different types of assets. Thus, there is also an aggregation problem.

Starting first with the problem of how to estimate the *ICT capital stock* (or the capital stock of any of the three major types, we assume that there exists a sufficiently long time series of investments in ICT at current prices and a suitable corresponding price index. If we let the current price investment for ICT asset type i in year τ be $I_{t-\tau}^i$, and the relevant price index be $p_{t,0}^i$, the productive stock of the actual type of ICT asset K_t^i at the beginning of period t can be computed as:

$$K_{t}^{i} = \sum_{\tau=0}^{T^{i}} \left(I_{t-\tau}^{i} / p_{t,0}^{i} \right) h_{\tau}^{i} R_{\tau}^{i}$$
(2.1)

where T^i represents the maximum service life of asset i^7 , h^i_{τ} is an age efficiency function representing the lower efficiency of older vintages of ICT capital goods, and R^i_{τ} describes the probability of survival of capital goods over a cohort's life span.

This becomes extra problematic at the sectoral level since ICT equipment might be rented and ICT services out-sourced.

⁵ For an overview of the problems measuring ICT prices, see Ahmed, Schreyer & Wölfl (2004).

In OECD work the average service life for different types of ICT capital is assumed to be 3 years for software, 7 years for ICT hardware and 15 years for communication equipment (Schreyer, Bignon & Dupont,

Turning now to the *aggregation* problem, we need a means to effectively incorporate differences in the productive contribution of heterogeneous investments as the composition of investment and capital changes over time. User cost weights provide such a means since, under assumptions of competitive markets and equilibrium conditions, they reflect the marginal productivity of different assets (Jorgensen, 1963; Jorgensen & Griliches, 1967). User costs are imputed prices and reflect how much would be charged in a well-functioning market for a one period-rental of a given capital good. Ignoring tax effects, the user costs of a capital good i, u_t^i , are composed of (i) the net rate of return r_t applied to the purchase price of a new capital good p_t^i , (ii) the costs of depreciation, captured by the rate of depreciation d_t^i , and (iii) the rate of change of the price of the actual capital good expressed by $\psi_t^i \equiv d \ln p_t^i$:

$$u_{t}^{i} = p_{t}^{i} \left(r_{t} + d_{t}^{i} - \psi_{t}^{i} \right) \tag{2.2}$$

The expression in the parenthesis represents the gross rate of return on an investment in a new capital good in year t. The gross rate of return on investments in infrastructure capital goods tends to be higher than for other types of capital goods. This is a result of the rapid obsolescence of ICT capital goods, which enters the user cost of capital in the form of purchase prices of new capital goods and via the rate of depreciation. Falling purchasing prices makes it less expensive to buy new capital goods but raises the costs of holding old capital goods. Depreciation rates may be computed according to different formula but they all shall reflect the relative loss of the value of a capital good due to ageing.

2003). Of course, these assumptions are critical and one can wonder whether the assumed average service lives might be too long, given the rapid technological changes in the field of ICT.

3 The Economic Impact of ICT Diffusion

We have in recent years been able to observe what can be described as an evolutionary process whereby economies at the national and the regional level and all their sectors are being transformed by the rapid development, adoption, and use of ICT innovations. In this respect, ICT functions as a *new generic general purpose technology*, which impacts these economies both broadly and deeply by generating a wide array of new products, production processes and services (Brynjolfsson & Kahin, 2000; Mowery & Simcoe, 2002). Carlsson (2003) takes this idea one step further arguing that ICT, which involve among other things a combination of digitalisation and the Internet, seem to have broader applicability than previous general-purpose technologies. It not only affects all manufacturing industries but also, and even more so, all different service industries, which account for an increasing and dominating share of the economy in developed economies. Furthermore, it has given rise to new industries within both the manufacturing and the service sector. However, one should observe that it is a common feature of new general purpose-technologies that it takes a long time before they are implemented (including the necessary organisational changes) and used in such a way that they could develop their abilities to the fullest (David, 1991).

It is unquestionable that the effects of the development, spread, and use of ICT go much further than changing the industrial composition of developed economies. ICT are playing an increasing role in economic growth, capital investments, and other aspects of the macroeconomy (Brynjolfson & Kahin, 2000). The emergence of new goods and services as well as changes in the characteristics of old goods and services due to the use of ICT, including the ways good and services are produced and distributed lead to changes in market structures and competitive conditions affecting and creating new opportunities for small firms and entrepreneurs. As ICT are routinely deployed in organisations to re-engineer processes, gain new strategic advantages, or network across organisational boundaries, they change both the internal organisation of companies and other organisations and the relationships between companies and organisations (OECD, 2002). The adoption of ICT allows for a reduction of transaction costs and leads possibly to more efficient markets (Malone, Yates & Benjamin, 1987; Lee & Clark, 1997).

ICT are a genuine source and generator of new business models and new wealth, but they are also undermining old business models and threatening and even destroying investments and

jobs in certain established businesses. In addition, the spread of ICT is changing the labour market by generating *new ICT-occupations* and at the same time changing the requirements for non-ICT jobs. Due to the structural changes generated by ICT, employment is increasing in some sectors and declining in others.

Since the conditions to develop and to use ICT vary substantially between regions, we expect large variations between regions in the timing as well as in the extent that they are affected by ICT.

A common view is that ICT have a large impact on production and business processes and that they thus are a *major stimuli of economic growth*. However, there seems to be substantial disagreement about the form of this impact and researchers in the field seem mostly to use one of two major approaches (Smith, 2002).

The first approach argues that economic growth is driven by the *emergence of new sectors* embodying *new technologies* including the ICT-producing sectors themselves. In this case growth comes from two sources: (i) new sectors exhibit higher growth rates of value added, productivity and incomes and will thus function as a source of growth for the whole economy, and (ii) new sectors change the conditions of other sectors of the economy by changing relative prices, and by providing a new set of inputs that raises productivity either by the introduction of new or improved products or new production methods. The production of ICT and the emergence of new ICT-based industries contribute directly to increase GDP and to boost aggregate productivity.

The second approach argues that, since ICT represent a special type of capital good, increased investments in ICT by companies and governments will raise labour and total factor productivity. Investments in ICT complement or replace investments in other capital goods and increase the capacity of the production of ICT-using sectors and industries.

However, there is a third potential indirect growth impact – *spillover effects*. When the spillover effects of technological advances from industries producing ICT to industries using ICT takes place, *an increase in total factor productivity* can be achieved (Jorgensen, Ho & Stiroh, 2002; van Ark, 2002). It must be observed that the benefits of investing in and using ICT depend on *sector-specific effects*. ICT are more important to raising productivity in certain sectors than in others and since different countries have different sectoral

specialisation, their gains from investing in and using ICT will differ. Investments in and use of *communication network technologies* present a special case because of the benefits derived from spillover and network effects (OECD, 2003; Dederick, Gurbaxani & Kreamer, 2003; Meijers, 2004).

3.1 ICT and Structural Change

Already in the 1980s researchers started to claim that ICT represent a sector and a technology with a potential to generate *structural* change in the world economy, i.e. the expected quantitative effects were large and the expected qualitative effects had the potential to generate a totally new type of economy (Freeman & Perez, 1988). A basic idea here is that economic growth in some sense is related to *qualitative changes* in the sense that the industrial, the occupational, the educational, etc., structure of the economy is changed. It can involve a more advanced division of labour as described by Adam Smith, which allows on the one hand increases in productivity and on the other hand a spin-off and out-sourcing of new activities (Stigler, 1951). Kutznets (1959) stressed that a high rate of growth depends upon a continuous emergence of new inventions and innovations, which provide the basis for new industries whose higher growth rate compensates for the slower growth rates among older industries.

The first to present a more systematic view of innovation as a driver of structural change was Schumpeter and in particular in his book *Business Cycles* from 1939. He does not offer any coherent theory of the generation of innovations but he stresses three points that he sees important:

- Innovations are clustered together and are not evenly distributed in time,
- Innovations concentrate in certain sectors and their surroundings,
- There are discrepancies between sectors: some industries move on, others stay behind.

The historical role of pervasive technologies, such as ICT, has been intensively discussed in the Schumpeterian literature on economic growth and structural change, which started to flourish in the 1970's. This literature presents a framework that explains the subsequent *rise* and fall of pervasive technological systems and their interaction with different sectors in the economy. What this framework suggests is that major technological breakthroughs, structural change, and economic growth are closely related, and can only be analysed jointly.

Freeman & Perez (1988) systemizes Schumpeter's work on Kondratieff waves and develops this into an argument that the key driving force of growth at this time is ICT. In their framework, economic growth is driven by *radical technological changes* that shift the entire 'techno-economic' paradigm involving (i) new forms of best-practice organisation, (ii) new skill profiles in the labour force, (iii) new location patterns, (iv) new infrastructures, (v) new consumption patterns, (vi) new types of dominant firms, etc. It is still unclear how these dramatic changes in economic life exactly are related to ICT. However, this approach is often used to create arguments for the role of ICT in modern times. Fagerberg, et al., (2000) argue that what matters for economic growth is the ability to exploit areas of *high technology opportunity*, which in recent decades have been dominated by ICT. Furthermore, they claim that their analysis show that Europe has lost ground in a number of strategically important sectors, particularly those related to ICT.

Verspagen (2004) uses a Schumpeterian framework to make a systematic analysis of the role of ICT in the structural change of the US economy over most of the post-war period. His aim is to relate the role of structural change in connection to a specific historical case of a major technological breakthrough, i.e. ICT. What he shows is that even if ICT have substantial effects on the structure of the economy, one can not draw the conclusion that ICT is the main pervasive technology of our days in generating *technology spillovers* or that it will substitute older technologies completely. The picture is rather that ICT is an *important complement* rather than a substitute to older technologies, which will continue to play an important role in the economy.

However, there are many problems associated with this approach and it is open to a number of quite basic objections (Smith, 2002):

- 1. There is a strong tendency to conflate innovation and diffusion and assuming that radical innovations generate rapid impacts. Technologies, such as ICT, take a long time to diffuse and an even longer time to have an impact.
- 2. It is not necessary that new sectors contribute to output in a significant way even when they are fully established. Hardware and software ICT industries are still rather small.

_

It is important to observe that structural changes also are a key feature at the micro scale. ICT redistributes/reallocates work tasks across persons, positions, and operations inside each organisation as well as between organisations. Successful implementation of an ICT project requires reorganisation of the firm around the new technology (Helpman & Trajtenberg, 1998, Yang & Brynjolffson, 2001; Brynjolfsson, Hitt & Yang, 2002).

3. This approach cannot explain growth in countries that does not possess a substantial ICT producing sector and many of these countries are high growth countries.

3.2 The Effects of ICT on Aggregate Productivity and Output Growth

Analysing economic growth economists have traditionally emphasised the *accumulation* of conventional inputs such as labour and production and infrastructure capital as main force behind output expansion. The basic theoretical background was laid by American economists who from the mid-1950's attempted to isolate the relative contributions of capital investment and technical change to labour productivity growth in the U.S. Solow (1957) was able to demonstrate that the long-run economic growth in the U.S. could not be explained by growth in labour or capital but was instead explained by what Solow termed "*technical change*" that actually was the unexplained residual. To try to disentangle the components of technical change a "growth accounting" research programme was set up in the US (Denison, 1962). The basic *neo-classical approach* applied by Solow and many of his followers consists of a growth equation that relates output to the level of technology – a technology shift parameter – and the inputs of capital and labour. This makes it possible to estimate the extent to which output grows independently of factor inputs, i.e. to estimate "technical change". By quantifying specific inputs, such as *investments in ICT*, it is possible to estimate its role for growth in labour productivity or total factor productivity.

More recently, economists have paid more attention to other growth stimulating factors such as the *sources of technological change* and *institutions*. Following the contributions of Romer (1986 & 1990), Lucas (1988), Grossman & Helpman (1991), and Aghion & Howitt (1992), numerous studies of economic growth place technological change at the heart of the growth process. This change of focus has been stimulated by theoretical achievements, which allow *micro-economic* aspects of the *innovation* process to be linked to macro-economic outcomes.

Many growth analysts have in recent decades tried to estimate the effects of ICT on economic growth. Since ICT represent a GPT, its *growth effects* has been analysed at the *macroeconomic* level as well as at the *sector* level. At the sector level, it is possible to distinguish between studies, which analyse the growth of *ICT producing sectors* and studies, which analyse the effects of the *investments in ICT in other sectors*. In other sectors, the effects of investments in ICT can manifest themselves in terms of higher productivity in firms

and sectors that invest intensively in ICT. Investments in ICT should also give higher average rates of returns than alternative types of investment in the economy.

One can identify two major approaches for estimating the *effects of investments in ICT* on economic growth:

- 1. Estimations of the *productivity effects* of the ICT capital stock based upon quantifications of this capital stock including both hardware and software (Oliner & Sichel, 1994; Sichel, 1997.)
- 2. Estimations of "technical change" with a production function approach and relating productivity growth to the use of ICT (Bailey & Gordon, 1988; Jorgensen & Stiroh, 1999).

Before trying to summarize some of the major empirical results, it is important to highlight some of the limitations of the empirical studies (Smith, 2002):

- Basing the econometric estimations on a production function approach implies the
 assumptions that (i) the economy in question is in some sort of competitive
 equilibrium, (ii) investment in ICT is rational, and (iii) investments in ICT earn a
 normal rate of return at the margin.
- The studies normally disregard the possibility that growth can come from other sources but new inputs and new knowledge and technologies embodied in them.
- Most studies concern the US economy, which makes generalisations about the effects
 of investments in ICT in other countries difficult.
- The studies vary a lot in terms of unit of analysis, performance concepts and measures, input measures and type of econometric analysis, which makes it difficult to compare them (Wilson, 1995).
- Data reliability can often be questioned.

Looking first at the US, the results of the empirical studies seem to have changed over time. Studies covering time-periods up till the early 1990s seldom report any significant impacts of ICT. Franke (1987) in a study of labour productivity in insurance and banking between 1958 and 1983 found declines in capital productivity associated with specific ICT innovations. A study of labour productivity in the service sector by Roach (1991) found large-scale increases in ICT capital stock relative to other capital inputs coupled with stagnant productivity suggesting no payoff from ICT. Morrison & Berndt (1991) and Berndt & Morrison (1995) in

a study of labour and total factor productivity in 20 manufacturing industries for the period 1968-1986 found that ICT capital has no impact on productivity compared to non-ICT capital.

Turning to studies that report significant impacts of ICT, we first have Siegel & Griliches (1991), who in a study of US manufacturing found significant impact of ICT on total factor productivity but who also expressed serious doubts over data reliability. Large returns on ICT capital investments and a disappearance of the productivity paradox was found by Brynjolfson & Hitt (1993) in a firm level study of US manufacturing for the period 1987-1991. More recently, a literature has emerged, which more unambiguously has claimed that investments in ICT have driven growth in the US economy since 1995. US productivity growth was at record levels during the period 1995-2000 and so were investments in ICT. Several authors have claimed that this is the long-expected payoff to investments in ICT and lean to the view that ICT have played a significant role in generating a fundamental change in the U.S. economy's growth (Oliner & Sichel, 2000; Jorgenson & Stiroh, 2000). Despite some methodological differences, these authors derive similar estimates:

- a high contribution of the ICT sector to growth in labour productivity, ranging from 41
 to 55 %;
- 2. around a quarter percentage point of the acceleration in labour productivity since 1995 is attributed to ICT (total factor productivity growth in the ICT sector);
- 3. industries with the highest investments in ICT also registered the highest increases in labour productivity (cf. Stiroh, 2002b);
- 4. along with the ICT-producing industries, ICT-using industries also played a fundamental role in accelerating productivity and growth, with service industries making a particularly important contribution to growth (Triplett & Bosworth, 2002), and ICT contributing a half a percentage point to capital deepening (all of which is attributable to the accumulation of ICT capital). In total, ICT seem to have contributed three-fourth of the labour productivity acceleration between 1995 and 2000 and the ICT-producing and the ICT-using sectors are estimated to have contributed 23 % and 25 %, respectively, to American economic growth (Jorgensen, 2001).

Gordon (2000) and Bosworth & Triplett (2000) represent a more critical view and claim that the ICT "revolution" has not had the same impact as the general-purpose technologies introduced in the past century (such as electricity or transportation). However, one should remember that the effects of general-purpose technology revolutions historically seem to

generally have occurred in three (often overlapping) main stages. First, technological change raises productivity growth in the innovating sector; second, falling prices encourage capital deepening; and, finally, there can be significant reorganisation of production around the capital goods that embody the new technology. Gordon (2000) focuses on the cyclical component of the US productivity surge, suggesting that half of the acceleration after 1995 was a cyclical phenomenon.

Turning now to studies at the macroeconomic level that include several European and other OECD countries we find that ICT have impacted macroeconomic variables here too, though in many countries on a lesser scale (Mairesse, Cette & Kocoglu, 2000; Daveri, 2002; Jalava & Pohjola, 2002; Coleccia & Schreyer, 2002; COM, 2003). It seems as if the contribution of the growth of ICT capital assets to GDP growth in the OECD countries in principle doubled from the period 1990-1995 to the period 1995-2001 from on average 0.25 to on average 0.50 percentage points (Ahmed, Schreyer & Wölfl, 2004; Jorgenson, 2001, Colecchia & Schreyer, 2001; van Ark, et al., 2003; OECD, 2003). In relative terms, the contribution of ICT capital assets to GDP growth seems to have increased from about 16 % of total GDP growth to about 20 % between the two periods. What is intriguing is the *large dispersion* in the contribution of ICT capital assets to economic growth in different OECD countries. Very strong contributions have been observed for the United States, Canada, the Netherlands and Australia amounting to about one fourth of GDP growth over the period 1995-2001. However, for other OECD countries, such as France, Finland, Portugal, and Germany the estimated contribution of ICT capital assets to economic growth is much smaller.

One important reason to the increased contribution of ICT capital assets to economic growth during the second period is the increased importance of ICT capital to growth in total capital input. While non-ICT capital contributed most to capital growth in the period 1990-1995, ICT capital contributed to between one third and half of total capital growth between 1995 and 2001 in most OECD countries (Ahmed, Schreyer & Wölfl, 2004). ICT hardware accounted for the largest share of the contribution of ICT capital to growth in total capital during the 1990s, but ICT software and ICT communications equipment seems to have become increasingly important.

In a panel study of 25 OECD countries, Belorgey, Lecat & Maury (2006) show that both production of and spending on ICT have a positive effect on the labour productivity growth

rate. They also find that average spending on ICT 1992-2000, used as a proxy for the contribution made by ICT has a positive effect on the level of labour productivity.

In Europe, in spite of a significant acceleration in investments in and use of ICT between 1998 and 2001, the productivity gains were limited and the disparities between countries wide. According to van Ark (2002), annual labour productivity growth between 1995 and 2000 was only 1.3 % per year. In recent years, a quarter of EU GDP growth and 40 % of its labour productivity growth is estimated to be due to investments in ICT, while 60 % of U.S. productivity growth is explained by these technologies (COM, 2005a). This implies that there within many countries within the European Union is a *potential for catch-up* in terms of investments in ICT capital assets.

What are then the reasons for the divergence between EU and the U.S. in terms of taking advantage of the benefits of investments in ICT? Actually, it is very intriguing that while the consumers in some countries in Western Europe have been quick to adopt new ICT products the overall picture is that Western Europe is lagging behind the US and Japan. Disregarding possible measurement errors, not least concerning the service sector, one explanation is the differences identified in the sectoral productivity structure between the two regions. First, the EU is not as highly specialised in ICT-producing sectors as the U.S. Second, ICT-using sectors in the EU have gained lower benefits with regard to total factor productivity growth compared to the same sectors in the U.S. (COM, 2005b). Except for a few small countries, such as Sweden and Finland, manufacturing and service industries in Western Europe have been much less alert at exploiting the potential for new markets and new ways of organizing production and distribution. It has been claimed that the most important difference between the US and Western Europe is that firms in Western Europe have failed to change the way they do business in response to the new technologies (Gordon, 2004). Actually, productivity growth in Western Europe compared to the US seems to have been particularly slow in three main ICT-using service sectors, namely, retail, distribution and financial services (O'Mahony & van Ark, 2003, Eds.). It is quite possible that the extent of market regulation in Western Europe significantly slows the speed of adoption of new technologies and new ways of doing business (Gust & Marquez, 2004). However, it should be noted that labour market regulations differ widely across the countries in Western Europe, which make generalizations difficult.

Van Ark & Piatkowski (2004) investigated the productivity performance of the 10 new EUmembers in Eastern Europe (CEE-10) and in EU-15 to detect sources of convergence between the two regions. They show that changes in labour intensity have been an important source of productivity convergence during the 1990s, and assume that it is likely to be so in the near future too. Their study also found that despite lower income levels, ICT capital in the CEE-10 has contributed as much to labour productivity growth as in the EU-15. Analyses of different industries show that manufacturing industries that have invested heavily in ICT have been a key factor in the restructuring process. As such ICT may therefore have been an important source of growth but probably a temporary source of convergence. In the longer run the impact of ICT on growth will have to come primarily from its productive use in services. The paper also includes a New Economy Indicator that reflects the existence of an economic environment conducive for continued investments in and use of ICT. It shows that further reforms are much needed for CEE countries to enter a second convergence phase in the coming decades.

3.3 The Effects of ICT on Productivity and Output Growth at Firm Level

There is a lack of empirical evidence about the relationship between investments in and use of ICT and productivity and output growth outside the U.S., mainly due to a lack of internationally comparable estimates of investments in and use of ICT capital at this level (van Ark, 2002; Devaraj & Kohli, 2000; Crowston & Myers, 2004). Much clearer and stronger evidence of the impacts of ICT comes from evidence at the firm (company) level (Bryjolfsson & Hitt, 1996; Baily & Solow, 2001; Brynjolfson, Hitt & Yang, 2002; Bresnahan, Brynjolfson & Hitt, 2002; COM, 2003; OECD, 2003; Kohli & Devaraj, 2003). While spillovers from ICT are typically not found at industry level (Stiroh, 2002 & 2003), there exist firm-level evidence that ICT in the US has a larger impact on productivity than suggested by its share of total costs (Brynjolfsson & Hitt, 2000 & 2003; OECD, 2004)

At the firm level, ICT influence production in several ways. First of all they enter production embodied in capital goods with their pertinent software (Jorgensen, 2001) which promote productivity not only by reducing production costs but, above all, improving the quality, the flexibility, the reliability, and so on, of processes and products. Second, the new capital goods make it possible to produce new types of goods and in particular new types of services. Third, ICT components make it possible to improve the quality of existing products as well as developing totally new products that are more highly valued by customers. An important element over time is the rapidly increasing capacity of ICT capital goods and components at the same time as their costs have decreased continuously.

The use of ICT in the production process increases labour productivity not only through automation and the transfer of tasks to customers, but also through capital deepening. As coordination technologies, ICT generate major impacts on firms because they allow for a more efficient use of information, which increases efficiency as well as give rise to synergies. The introduction of ICT improves the access to information within firms, thus enabling more effective and more rapid decision-making by employees and managers (OECD, 2003; Dederick, Gurbaxani & Kraemer, 2003). As firms have introduced ICT, they have also in a parallel process or as a result of the introduction of ICT changed their internal organisation, among other things to make the organisation more flexible. Flexibility has been enhanced by self-managed teams, multi-tasking, just-in-time production and delivery, total-quality management, and decentralised decision-making (Aubert, Caroli & Roger, 2006). A number of studies show that ICT and changes of the internal organisation of firms have significant positive effects on labour productivity (Black & Lynch, 2001; Brynjolfsson & Hitt, 2000; Bresnahan, Brynjolfsson & Hitt, 2002). In particular, it seems as if it is through their role as coordination technologies that ICT have a special impact on total factor productivity at the firm level (Brynjolfson & Hitt, 2000; Brynjolfson, Hitt & Yang, 2002; Dedrick, Gurbaxani & Kraemer, 2003).

With regard to the impact on the production process as a whole, the use of ICT improves the competitiveness of firms making it possible for them to increase their market share by becoming leaner than their competitors. The use of ICT also helps firms to expand their product ranges, customise the services they offer and/or respond better and quicker to customer demand. The use of ICT also makes it much easier for firms to *outsource* and even *offshore* many of its activities and instead concentrating on its core business and core competence. There are substantial evidences that disparities in aggregate productivity growth are mainly due to differences in performance at the industry and also at the firm level including the entry of high performing new firms (Cohen, Garibaldi & Scarpetta, 2004).

Results obtained in recent years show that while investments in ICT are necessary, they are not sufficient to guarantee that firms will achieve full productivity benefits (Lera-López & Billón-Currás, 2005). Full effects of investments in ICT in terms of use of ICT will be achieved only when accompanied by complementary capital investments, investments in

_

⁹ Bertschek & Kaiser (2004) criticize these studies for assuming that increases in labour productivity induce an internal reorganisation of firms. They present empirical evidence, which indicates that labour productivity and the internal reorganisation of firms are simultaneously determined.

human capital, changes in the organisation of firms and production systems, development of improved and new goods and services, and so on. However, changes within existing firms are not enough to get the full productivity benefits from investments in ICT. Of critical importance is also the institutional framework within which firms operate, which determines the conditions for investments and competition as well as for innovation and entrepreneurship. Lack of competition within service sector industries within the EU due to too-long remaining national regulations and protection is most probably one major reason why these industries within the EU has been lagging in terms of innovation and productivity compared to the same industries in the US. Looking at the regional level within the EU it is obvious that there are substantial variations in terms of formal institutions (laws, regulations, collective agreements, etc.), ¹⁰ as well as in informal institutions (norms, cultures, traditions, customs, practices, etc.), which influences the propensity to invest and adopt ICT as well as the effects of the diffusion of ICT.

¹⁰ Informal institutions are part of the wider concept "social capital" (Westlund, 2006). Obviously, regional variations in social capital have an influence on ICT investments and the effects of ICT investments.

4 Regional Dynamics and Economic Transformations during the Deployment Phase of ICT

During recent decades, production and innovation systems have going through radical but simultaneous and interwoven transformations due to new competitive strategies, globalisation, the emergence of the knowledge society and the ICT revolution. This has created quite new conditions for the business community as well as for the political community. However, the *geographical impacts* of these developments in general and of the ICT revolution in particular have been disputed. Some argue that we are heading for a world where businesses operate without considerations of supra-regional, national, and regional boundaries, where multinational companies act without any distinct home base and where ICT including the use of E-commerce and the Internet exclude time and space as important parameters. Others claim that geographical proximity increase in importance to businesses in order to be able to create inter-firm networks based on trust, reciprocity, and interactive learning. As we will try to show below, both stories are true to a certain extent and both processes are running simultaneously and in intense interaction with each other.

In most advanced economies, an ever increasing share of economic inputs and outputs is in the form of ICT and knowledge (Bristow, 2003). As a result, the traditional determinants of *industrial location* – access to raw materials, transportation networks, low costs, a large pool of general labour – are becoming less important for location within these economies. Instead, *locational choice is increasingly becoming governed by access to particular skills, technology, and knowledge, as well as entrepreneurial talent and venture capital.* Of particular importance is the provision of ICT skills, ICT technology, ICT knowledge, ICT services, ICT entrepreneurial talent, and ICT competent venture capital (Johansson, 2006).

Although, there is now a substantial body of literature on the spatial consequences of the increased use of ICT in the economy, much of it is inconclusive (Johansson, Karlsson & Stough, 2006). One reason might be that the context is rapidly changing not least due to the success of the Internet and e-commerce. Even if much interest have been devoted to the issue of how investments in ICT capital and the use of ICT induce *spatial transformations*, much less interest have been devoted to how these transformations affect regional economic growth. This is interesting per se since several economists have suggested an important link between national economic growth and the concentration of people and firms in large urban regions

(Karlsson & Johansson, 2006). The high concentration of people and firms in large urban regions creates an environment in which knowledge moves quickly from person to person and from firm to firm. This implies that large, dense locations encourage *knowledge diffusion* and exchange, thus facilitating the spread of new knowledge that underlies the creation and imitation of new products and new ways to produce products (Carlino, 2001).

The New Economic Geography (NEG) theory, which has developed since the early 1990s, provides theoretical tools to understand the *factors driving spatial transformations and the effects of these transformations on regional economic growth*. It starts with the presumption that *functional regions* and not countries are the natural units for economic analysis. The reason is that economic activities are not evenly distributed across space and show clear tendencies to agglomerate. The NEG theory explains why economic activities concentrate in certain regions and not in others (Krugman, 1991; Fujita, Krugman & Venables, 1999; Johansson, Karlsson & Stough, 2002, Eds.)

The increased use of ICT enables major reductions in geographical transaction costs by reducing spatial information frictions (Flamm, 1999; Sichel, 1997). Examining the interrelationships between three variables - increasing returns due to scale economies, demand for final products and geographical transaction costs – in a world with monopolistic competition makes it possible to draw some general analytical conclusions concerning the effects ICT-induced reductions of geographical transaction costs. When geographical transaction costs are reduced, producers in large regions, i.e. regions with large home markets, which already have good opportunities to exploit economies of scale due to a large home market, can lower the production costs by also delivering to other regions, i.e. by increasing their exports. When exports increase, there will also be increases in incomes, which induce more producers of differentiated products to start production in the large region. Increased exports also imply an increased demand for differentiated inputs, which will induce more producers with their internal scale economies to start producing such inputs. As a consequence, we have a situation with cumulative causation or positive feed-backs initiated by the effects of ICT on geographical transaction costs. Thus, as first conclusion we may assume that investments in ICT and particular in communications equipment stimulate further agglomeration.

The original Krugman version of the NEG theory has nothing to say about the role of knowledge in regional economic growth. Increasing returns is the result of the exploitation of economics of scale in production only. However, since the development and exploitation of ICT is intimately associated with the development, diffusion, appropriation and use of knowledge it is necessary to integrate knowledge and knowledge externalities in the above framework. The *literature on innovation systems* strongly indicates that *knowledge flows*, *including spillovers are at the core of regional development* (Karlsson & Johansson, 2006). Since knowledge sources have been found to be geographically concentrated (Audretsch & Feldman, 1996), location is crucial in understanding knowledge flows (Karlsson & Andersson, 2007; Andersson, Gråsjö & Karlsson, 2007). In addition, the capacity to absorb flows of new knowledge is facilitated by geographical proximity (Jaffe, Trajtenberg & Henderson, 1993; Baptista & Swann, 1998). Already Marshall (1920) identified the exchange of ideas as a type of externality leading to localisation, i.e. clustering, of economic activities.

Large, dense regions offer special advantages in terms of knowledge flows and knowledge spillovers, since they combine the localisation of clusters in specific industries with industrial diversity, i.e. with a range of different industrial clusters. This suggests a formulation of a NEG model based upon knowledge externalities. When a (large) functional region has achieved an initial advantage in knowledge production due to e.g. a large pool of well-educated labour and a rich supply of ICT capital assets, it will attract (i) knowledge-creating and knowledge-utilising firms, since it offers opportunities to take advantage of increasing returns in knowledge production and knowledge use including imitation, and (ii) knowledge-rich labour, which wants to take advantage of the increasing demand for its skills. With increased knowledge intensity in larger regions we can expect increased investments in ICT capital assets, which will further reduce geographical transaction costs.

4.1 ICT and Spatial Transformations

The claim above that the interaction between knowledge and ICT will stimulate further agglomeration clashes with the predictions of some cyber prophets and technological optimists. They have claimed that the emergence of the digital economy would kill distance and make urban regions superfluous (Friedman, 2005; Cairncross, 1997; Knoke, 1996; Naisbit, 1995; Negroponte, 1995; Toffler, 1980), and at the same time eliminating the scale disadvantages of smaller and more peripheral regions. Their basic idea was that the spread of the use of ICT has the potential to replace face-to-face activities, i.e. to substitute physical movements that formerly occurred in central locations, which would strongly reduce or even eliminate agglomeration economies and hence make all economic activities totally "foot-

loose". These deterministic views see developments and investments in ICT as radically reshaping society, and by extension, cities.

However, the difficulties in forecasting the future spatial and social impact of ICT is illustrated by Salomon (1998), who demonstrates the *complexity* by reviewing the case of telecommuting as a travel substitute. In his study, he stresses that technologies are social constructs and thus, in order to forecast the impact of such technologies, the way the individual decision-maker penetrates such a technology must be understood, as well as the extent to which individuals (and firms) adopt it and change their behaviour accordingly. Nevertheless, the relationship between transport technology and ICT has received much interest in recent decades (Salomon, 1986; Nilles, 1988; Mokhtarian, 1991; Hepworth & Ducatel, 1992; Mokhtarian & Salomon, 2001). There are many obvious reasons for this. Both technologies belong to the class of "friction-reducing technologies", both have a network structure, and there is, in some cases, a (probably overstated) potential for substitution between physical travel and virtual travel. As both technologies facilitate remote activities, there has been much interest in this potential substitution (Garrison & Deakin, 1988; Boghani, Kimble & Spencer, 1991). However, Mokhtarian & Meenaksisundaran (1999) remind us that alongside substitution effects between transportation and ICT, there is considerable evidence suggesting stimulation or generation effects as well, i.e. ICT can stimulate more physical travel and transport. Moreover, ICT can change travel and transport behaviour, not just the decision about the travel or the transport itself. ICT also offer tools to increase the quality of transportation networks and services.

In the literature, it is argued that *ICT open new complementarities and potential synergies*, which are most evident in the way ICT networks are becoming integral to an increasing array of traffic and transport operations (Giannopoulus & Gillespie, 1993, Eds.; Nijkamp, Pepping & Banister, 1995). Through better monitoring with the help of ICT, a better, faster, and timelier flow of goods and persons from their origin to their place of destination can in principle be realised. ICT is in this sense first and foremost a complementary technology to existing distribution and transportation networks. While the term e-commerce seems to imply a process of substitution of physical commerce, ICT is rather likely to increase the efficiency of the distribution and transport delivery systems through reduction in transport costs and better usage of transport infrastructure whether by ship, rail, road, or air transport.

Substitution might occur but rather between different, alternative transport infrastructure systems (Soete, 2006).

Today it seems clear that the "death of distance" picture is at least single-sided. As ICT have been adopted for decades (and if we include the telephone for more than a century), most researchers today seem convinced that cities are not going to disappear (Cohen-Blankshtain & Nijkamp, 2004). Graham and Marvin (2000) stress that most applications of ICT are largely metropolitan phenomena and that ICT and large metropolitan are mutually supportive phenomena. Not least, the development of new technologies and new products seems likely to remain grounded in the large urban regions in the advanced countries, which imply that these regions will keep their locational attractiveness. There is also increasing evidence that increased investments in and use of ICT actually reinforces the position of large cities and not least the leading urban regions (Castells, 1989 & 1996; Moss, 1991; Hall, 1998; Wheeler, Aoyama & Warf, 2000, Eds.). Kolko (1999) suggests that ICT have led to the "death of distance", but not to the "death of cities". However, Graham (2002) claims that both distance and cities are far from being dead, and that geography still matters (cf. Nijkamp, Linders & de Groot, 2002). Beyers (2000) accentuates that ICT may enable living far from the city, but he also argues that not only are many businesses in the information society strongly tied to localised markets, but it is also in urban areas that the people working in these sectors want to live, for reasons related to consumption and tastes, and dictated by spousal relationships and other social relationships.

Bellini et al (2003) examine the impact of ICT on the location patterns of industries in Italy and find evidence for increasing convergence of industrial structure across regions in line with the "death of distance" hypothesis. However, they also find that knowledge-intensive industries tend to cluster together suggesting that knowledge-intensity acts as a counterbalancing force to the dispersion effect of ICT.

Investments in ICT may not necessarily encourage the dispersion of economic activities due to the network and technology effects of the supply of ICT infrastructure (Ogawa, 2000). Grant & Bergiust (2000) argue that ICT networks will play the same role in the twenty-first century that streets and highways played in the twentieth century, since they both are "spatial technologies" (Couclelis, 1994). Just as the car affected the shape of urban regions, there is an expectation that ICT will change the cities. The "information highway" now inherits the role

of physical highways. Therefore, while transport was the "maker and breaker of cities" (Clark, 1957), ICT are now expected to inherit or share this role.

Already in the early 1990s, Goddard (1991) developed a conceptual model to assess the possible effects of investments in ICT on the urban form, which emphasizes the *effects of ICT on organisations*. He identifies three levels of analysis that are needed to evaluate the expected future effects: (i) the effects on the organisational level, (ii) the effects on infrastructure, and (iii) the effects on different sectors. Often much of the research about the effects of ICT on the urban form is concentrated on one single channel of research. In most cases there is no aggregate analysis that examines the overall and interrelated effects of these technologies on the city on the whole (and on the system of cities). Thus, the empirical evidences are eclectic and there is still no integrated picture of foreseeable changes.

The large urban regions in the advanced countries are concentrations of knowledge – human capital, universities and R&D activities – and *knowledge constitutes a critical input for productivity, economic growth and development*. These regions are also leading centres of innovation and imitation. Desrochers (1997) points out the importance of geographical location for the transmission of tacit knowledge and innovations between competitors, suppliers, and customers via face-to-face interactions. Cities are a means of reducing the fixed travel costs involved in face-to-face interactions. Even if in principle improvements in ICT could eliminate the demand for face-to-face interactions and make cities obsolete in this respect, empirical results point in the direction that the use of mediated contacts is mainly a complement to face-to-face interactions (Gaspar & Glaeser, 1998). The conclusion is that as ICT improve, the demand for interactions of all varieties, including face-to-face interactions, should rise. Furthermore, these regions are the home for new propulsive and emergent growth sectors such as tourism and cultural industries based upon face-to-face interaction (Andersson & Andersson, 2006).

It should in this context be observed that *the provision of network infrastructures vary substantially* making only certain locations viable for communication intensive organisations and activities. Thus, it should be no surprise that the majority of the firms in the Internet industry is concentrated in key metropolitan regions (Bristow, 2003; Zook, 2002) and that the same general pattern prevails for both the so-called Internet 'backbones' in the United States (Malecki & Gorman, 2001) and the multimedia industry. Interestingly, Zook (2000) shows that over time there seems to be a stronger connection between Internet content and the

information-intensive industries than between Internet content and the industries providing the computer and telecommunications technology necessary or the Internet to operate. Even if these agglomerations interact digitally over long distance, their existence does not suggest a geography of general dispersion or that the industries in question should be indifferent to distance or proximity (Leamer & Storper, 2001). On the contrary, these industries are heavily concentrated in existing large agglomerations, thereby at least in the short run reinforcing existing patterns of uneven development.

4.2 ICT and the Location of Firms

Over time, we expect ICT to affect patterns of concentration and convergence of industries. Concentration is the tendency of an industry to cluster geographically, while convergence is the tendency of an industry to become more uniformly distributed geographically. Traxler & Luger (2000) illustrate the complicated and multidimensional effects of ICT on firm location. In their study, they examined possible spatial effects of these new technologies on the location of firms and concluded that ICT can have two opposite effects: dispersion and reinforcement of concentration. Indeed a relatively large body of literature comes up with such contradictory conclusions about the expected effects of ICT, emphasising the complex effects of these technologies on the behaviour of people. Kolko (2002) found that ICT intensive industries exhibit slower convergence, i.e. deconcentration, than other industries. This result indicate that clusters of ICT intensive industries persist not because they are ICT intensive per se, but because they tend to rely on highly skilled labour.

However, the effects of ICT go much further than to the ICT intensive industries. Investments in and the use of ICT have had a very strong effect on trends, that started well before the general diffusion of ICT. In recent decades, we have witnessed a gradual denationalisation of in particular large companies. Internationalisation and globalisation of production and markets have created the preconditions for *locational choices based upon global rather than national considerations*. The possibilities for companies to move their activities within and between countries have increased considerably. A clear tendency is that company units and plants are located where the conditions are the best whether we are talking about R&D or the production of standardised components. It is in particular two technological conditions, which have made the new scenarios possible. They are *production decomposition* and *network control*. Production decomposition implies that the production of a certain product can be

divided into separate stages that take place in different production units. To keep such a production system running there is a need for network control.

These new scenarios involve increased outsourcing as well as increased off-shoring of production. Famous historical examples of this is the production of semi-conductors where production is globally decomposed and involves multiple locations in several countries and the value chain controlled in an integrated way, often by multinational enterprises but also the production of products, such as mobile phones, computers, cars and airplanes. The degree of off-shoring is among other things a function of how easy it is to decompose a production process into different stages, and the labour-intensity of the intermediate production steps (Grunwald & Flamm, 1985) While outsourcing and off-shoring of production has been present in several decades, what is much more recent is the outsourcing and, in particular, the off-shoring of service production, which have become possible due to technological advances in ICT, declining real prices of ICT, large investments in ICT network infrastructures, e.g. broadband and mobile phone networks, rapid increases of telecommunication connections, including broadband connections, and decreased costs for air travelling (Kirkegaard, 2004b). These developments have made services increasingly tradeable (ICT-enabled services) and reduced the constraints on the choice of location for the production of services (Friedman, 2005; Abramovsky & Griffith, 2005).

Technical progress has reduced the optimal scale for a large number of economic activities. This implies that many small production units can replace a large production unit without efficiency and productivity losses. Thus, it has become possible to divide the production of goods and services between several or even many separated, local production and control units. In this case production is decomposed in a production chain, where several, separated production units each produces different components, while others take care of assembly, distribution and administration. However, a precondition is that production and flows of goods are controlled by means of ICT applications.

The degrees of freedom regarding the choice of location have increased manifold due to improvements and cost reductions within freight transportation, air travelling and ICT. The use of ICT and, in particular, the Internet makes it possible for companies to have *frequent interactions with suppliers, customers and their own production units* without daily face-to-face interaction, as long as the interactions concern routine contacts and standardised, well coordinated information flows.

4.3 ICT and Regional Economic Growth

Having discussed the relationship between ICT and regional transformations above it is now time to turn to the relationship between ICT and regional economic growth. One of the most stylized facts about economic growth is that productivity growth, rather than factor accumulation, accounts for most of the growth differentials across countries. Easterly & Levine (2001) argue that in the search for the secrets of long-run economic growth, a high priority should go to rigorously defining total factor productivity (TFP), empirically dissecting it, and identifying the policies and institutions most conducive to its growth. Even if the inflow of labour can play a somewhat larger role for economic growth at the regional than at the national level, we have strong reasons to believe that the major effect of ICT on economic growth goes via its effect on TFP in ICT-producing as well as ICT-using industries.

However, the development of ICT as well as the application of ICT seem to be critically dependent upon the availability of human capital in general and human capital with ICTcompetence, in particular. We can look upon ICT and human capital as complementary factors, Hence, we start our discussion of the relationship between ICT and regional economic growth from a theoretical perspective according to which the underlying source of sustained growth in per capita income, namely the accumulation of knowledge is endogenised through formal education, on-the-job training, basic and applied research, learning-by-doing, and process and product innovations (Aghion & Howitt, 1992), which implies that the indigenous innovative activities of regions become critical. This approach fully incorporates the Schumpeterian view of innovation as a result of deliberate efforts. New knowledge is not pure public goods, since even if it is non-rivalrous it is at least partly excludable. It is produced using existing knowledge and human capital through investments in R&D, which are renumerated by the temporary extra rent provided by the (partial and at least temporary) appropriability of the results of innovation in markets characterised by monopolistic competition (Romer, 1990; Grossman & Helpman, 1991). However, the existing accessible pool of knowledge increases because the benefits of generating new knowledge are not fully appropriated by the innovating firm due to knowledge spillovers, which benefits other firms in their innovative activities. Knowledge is a special type of "product", since it is not exhausted after use. Instead, it is cumulative by being based on the existing pool of knowledge.

This theoretical perspective, which includes innovative efforts into the determinants of growth, allows for permanent disparities in regional growth rates. Regions that are well-endowed in terms of knowledge and capital, due to their accumulated pool of knowledge will have a continuous advantage over regions less well endowed. The reason is that knowledge consists of organised or structured information that is difficult to codify and interpret, generally due to its intrinsic indivisibility (Karlsson & Johansson, 2006). As a consequence, knowledge is difficult to transfer without direct face-to-face interaction. This implies that proximity matters for knowledge transfer. Thus, knowledge flows much faster within than between regions. Even if ICT to a certain extent may change the conditions for knowledge flows, it is by no means given that this helps the less well-endowed regions. One can argue that the more well-endowed regions are in a better position to take advantage of the possibilities offered by ICT.

In a context like this, it is crucial to understand how knowledge is transferred between as well as within regions as well as among the actors involved. How is knowledge transferred between its source and its potential users? Starting with *inter-regional knowledge transfers* it is obvious that *multinational firms play a critical role*. Their intra-firm knowledge networks that also include the mobility of staff between different regions provide major links for knowledge transfers. Besides these links, embodied knowledge, which is the most critical part of knowledge, is transferred mainly via the mobility of knowledgeable people and capital goods including software. Turning to *intra-regional knowledge transfers* much evidence points in the direction the mobility of knowledgeable people and direct face-to-face interaction between such people are the most important channels for intra-regional knowledge transfers.

With knowledge spillovers given such a central role in the growth process, it is natural to ask which regional economic milieus are most conducive to knowledge spillovers? Does the specific mix of economic activities undertaken within any particular region matter (Feldman & Audretsch, 1999)? Glaeser et al. (1992) consider the factors that influence innovative activities in urban regions, and identify two relevant models in the economics literature. The first model, the so-called Marshall-Arrow-Romer model formalises the insight that the concentration of a particular industry within a specific urban region (Lösch, 1954) promotes intra-regional knowledge spillovers across firms and therefore stimulates innovation in that

particular industry. The basic assumption here is that knowledge spillovers mainly take place across firms within the same industry.

The second model regards *inter-industry spillovers* as the most important source of new knowledge. Specifically, Jacobs (1969) argues that the *agglomeration of firms in urban regions fosters innovation due to the diversity of knowledge sources located in such regions*. Thus, the variety of industries within an urban region can be a powerful engine of growth for that region, and the exchange of complementary knowledge across diverse firms and economic agents leads to increasing returns to new knowledge.

Given the relative importance of the two specialisation mechanisms, different regions may exhibit different growth experiences given their historically given economic structure. Given that ICT represent a general purpose technology and that the development of ICT is strongly concentrated to a limited number of urban regions a critical question is to what extent different regions offer good opportunities for knowledge related to ICT and the use of ICT to penetrate ICT-using sectors and industries, which in principle are all sectors and all industries represented?

Obviously, there are several factors that have to be accounted for to understand the regional growth, since regions have different capabilities to absorb and to transform accessible knowledge into (endogenous) economic growth (Crescenzi, 2005). It seems, for example, that the ability of regions to adopt and to adapt new technologies depends on the institutional infrastructure, education, geography, and resources devoted to R&D (Maurseth & Verspagen, 1999). These and other factors that influence innovation form a system of innovation, i.e. the network of institutions in the public and the private sector whose activities and interactions initiate, import, modify, and diffuse new technologies (Freeman, 1987). The systems approach is not a theory but a focusing device for identifying factors relevant for the innovation process (Edquist, 1997). Systems of innovation can be identified at the national level (Lundvall, 1992) but here we concentrate on regional systems of innovation (Andersson & Karlsson, 2006; Andersson & Karlsson, 2004), which exist as self-consistent and self-organised systems within the national ones (Howells, 1999).

Regional innovation systems can be seen as key building blocks and the engine in the innovative process. The process of innovation is still in a general sense governed by the national system of innovation but it is localised and embedded in a regional innovation

system. These regional innovation systems should be understood in terms of relationships and interactions between the various economic actors that make up the innovation system (Cooke, 1997), i.e. the innovative milieu (Camagni, 1995), where probably most actors are located in the region in question but others located in other regions nationally as well as abroad and integrated via various forms of network configurations.

Trying to understand the role of ICT for regional growth it is also important to acknowledge that ICT is nothing constant but instead in continuous change. Being a general purpose technology it changes over time as a result of scientific and technological advances, which increases its potential applications as well as reduces its costs, but also due to changes in the selection environment, which contribute to determine the timing and type of uses of the new technology. The selection environment is made up by all non-technological factors such as markets, supply of labour with the relevant training, infrastructure investments, institutional factors, and government regulation that to a varying degree affect the R&D carried out in the field, the kind of innovations launched and the speed of adoption of these innovations. However, despite powerful influences from the selection environment, ICT has rules and a momentum of its own, which determine the direction of how the technology develops. This implies that certain regions that have specialised in certain types of ICT might find that they are on the wrong trajectory as technology continuous to develop. The specialisation in mainframe and mini-computers in certain regions are obvious examples.

The effect of ICT on regional economic growth does come from two sources: the involvement of each region in ICT production and the speed of adoption of ICT in each region. Since, the involvement of different regions in the development and production of ICT as well as their selection environments for the adoption of ICT differ a lot, we shall naturally expect different effects of the diffusion of ICT on economic growth in different regions.

The extent to which different regions are involved in ICT production depends among other things upon historical initiatives by industry and/or national and/or regional governments and the past success of these initiatives. In those regions where the right conditions have prevailed ICT producing clusters have emerged based upon innovation, imitation and often the development of backward and/or forward linkages. Not least has technological imitation within different ICT industries played an important role in many successful ICT clusters, since such imitation is coupled with further technological innovation both by the imitating firms and by those firms whose innovations are subject to imitation. A critical factor for such

dynamic processes to evolve is of course that knowledge to a substantial degree can spillover between the firms involved.

Technological imitation stands for the inter-firm diffusion of innovations, i.e. for what might be called production or supply-side diffusion of innovations. The speed of inter-firm diffusion and the path of investments in production capacity by the firms in the industry is one factor determining regional economic growth. To the extent that the region is the market for the innovation the capacity growth and the competition between the suppliers will be one of the factors determining the speed by which the innovation is diffused in the region.

In terms of ICT production it seems as if regions can take advantage of ICT without being producers of ICT hardware, i.e. production of ICT hardware is not a necessary condition for ICT to have an effect on regional economic growth. Probably, the same prevails for ICT software in many cases. Furthermore, investments in ICT network infrastructure by different regions play an important role for the regional growth effects of ICT. However, due to the network structure of ICT infrastructures effects of investments in ICT infrastructure in one region may benefit other regions as well.

General purpose technologies, such as ICT undergo uncountable transformations over time. Naturally, suppliers invest resources to provide successively better and better versions of the products embodying ICT. These ongoing innovative activities within the ICT sector are yielding series of incremental improvements in existing ICT products at the same time as totally new ICT-products are developed. On the user side and here we focus on firms as users of ICT a similar process proceeds because as each user firm use a new piece of ICT to its production process or as an input in its products, it tends to make qualitative and quantitative changes in equipment, and to refine or add new features to its products as well as develop new products. Thus, ICT is used for product and process development in different user industries, in principle, in all industries including the ICT industry itself. Product and process development may involve the introduction of totally new products and processes, respectively, as well s the renewal of old products and processes. In terms of traditional production theory, product and process development based upon ICT give rise to new production functions.

Product and process development based upon ICT must be seen as part of the competitive strategy for a firm in a given industry. Every firm within an industry occupies a specific place

in the competitive spectrum. Its strengths and weaknesses with respect to particular products, to particular markets or vis-à-vis particular competitors will influence its choice of competitive strategies. Actually, product and process develop, together with sales or market promotion can be viewed as the major instruments for firms in the competitive struggle. In this competition, firms have three major strategic options: i) innovative competition based upon product development, ii) price or cost competition based upon process development, and iii) marketing competition based upon sales or market promotion. From a regional point of view it is obvious that the growth effect of ICT is much dependent upon the ability and the willingness of the firms in the region to adopt and implement ICT as a strategic competitive tool.

The potential of firms to use ICT to develop new products is depending upon the accessible market potential in different regions. The total market potential of a region consists of its own market potential and the accessible market potential in other regions. How accessible markets in other regions are depends upon the geographical transaction costs of different products. Obviously, firms in larger regions have an advantage when it comes to develop new ICT-using products due to a larger market potential – the home market effect. The potentials of smaller regions mainly are to be found in hardware and possibly software production given that they fully can take advantage of location economies by developing strong enough clusters. Another niche for smaller regions given that their ICT network infrastructure is good enough is to specialise on different types of call-centre activities. However, the development and supply of more advanced ICT-based services seems mainly to be a prerogative for larger regions, which have a large enough supply of qualified labour and a large enough supply of qualified customers, since the development and the supply of such services is critically dependent upon often frequent face-to-face interaction.

The adoption of innovations by firms at the regional level is on the one hand dependent upon their characteristics and on the other hand on the regional selection environment including the regional economic milieu offered by the actual region. Important firm characteristics are: i) size of firms, ii) economic and financial characteristics of firms, iii) the human capital characteristics of firms including the characteristics of their management, and iv) the internal and external communication networks of firms. The regional economic milieu is made up by among other things i) accessibility to regional and interregional market potential, ii) availability of production factors, and in particular, regionally "trapped" factors, such as

accessibility to educated labour, iii) the existence of external economies of scale in the form of localisation and urbanisation economies, and iv) the institutional framework including regional policies to stimulate innovation and innovation adoption as well as the social capital in the region. This implies that transport and ICT infrastructure is important since they are factors determining the prevailing accessibilities.

It is important to observe that there exists an optimal rate of innovation adoption and innovation diffusion, which implies that all new innovations should not be adopted immediately by all potential user firms. Due to the fact that many firms recently have adopted earlier varieties of ICT inputs for their products and/or ICT capital goods for their production processes, it is quite rational for them not to adopt every new potentially useful innovation immediately. Due to their recent investments these firms have a sunk-cost advantage of postponing an adoption until the sunk-cost advantage has vanished. This implies that regional policies aiming at stimulating the adoption of ICT innovations to stimulate regional economic growth must consider what is rational from the potential user firms' point of view.

Of course, there can exist and probably exist various market failures that might motivate certain regional policy initiatives to stimulate the adoption of ICT innovations by firms. One might here mention lack of information about new ICT innovations and their potential, lack of labour with the right ICT qualifications, the existence of unexploited positive external economies including learning economies, lack of ICT network infrastructure, etc. However, it is important also to consider the potential problems and costs of regional ICT policy in terms of the long time lags involved before policies have effects, the existence of asymmetric information, the lack of detailed information and knowledge about ICT among policy makers, the existence of vested interest in industry as well as among policy makers, the risks of distorting the function of markets, etc. Given this situation the best regional ICT policy to stimulate regional economic growth may in many cases be rather to improve the general economic milieu in the region in terms of transport and ICT network infrastructure, regional institutions and higher education including ICT education than to try to directly influence the ICT adoption decisions by firms.

5 Regional Innovation Systems and the Layers of Innovation

5.1 Regional Innovation Systems: Key Characteristics and Dimensions

In the past decade the innovation system (IS) approach has substantially enhanced our understanding of the nature of the innovation process, highlighting that innovation is an evolutionary, non-linear and interactive endeavour that requires intensive cooperation between firms and other organisations (Edquist 1997, 2005). Furthermore, inspired by the institutionalist school of thought (Hodgson 1988, 1999), the IS literature emphasises the impact of formal (laws, rules, etc.) and informal institutions (habits, routines, established practices, etc.) on innovation activities (Nelson and Winter 1982; Johnson 1992; Edquist and Johnson 1997; Edquist 2005).

Initially, the concept of innovation systems has been applied to the national level (Lundvall 1992; Nelson 1993; OECD 1999; Groenewegen and van der Steen 2006; Lundvall 2007). The literature on national innovation systems (NIS) has shown that countries differ enormously with respect to their economic structures, R&D bases, institutional set-ups and, consequently, innovation performances (Edquist 2001). Nations, however, can exhibit huge disparities in innovation across regions. This insight has provoked a growing interest by academic scholars in regional innovation systems (RIS). There are several reasons underscoring the relevance of the regional level as an adequate unit of analysis for studying innovation. First, there are marked differences between regions regarding their pattern of industrial specialisation and innovation performance (Howells 1999; Breschi 2000; Paci and Usai 2000, Hollanders 2007). Second, knowledge spillovers, which are ascribed to play a crucial role in the innovation process, are often spatially bounded (Jaffe 1989; Jaffe et al. 1993; Audretsch and Feldman 1996; Anselin et al. 1997; Bottazzi and Peri 2003). Third, notwithstanding increasing codification tendencies of knowledge (David and Foray 2003) tacit knowledge (Polanyi 1966) remains important for successfully carrying out innovation

_

In the 1990s also "non-territorial" specifications of innovation systems emerged, including technological innovation systems (Carlsson 1994) and sectoral innovation systems (Breschi and Malerba 1997; Malerba 2002). The scholars favouring the technological approach argue that systemic interrelationships are unique to technology fields. The protagonists of the sectoral approach examine how groups of firms develop and manufacture products of a specific sector and how they generate and utilise the technologies of that sector.

See, for example, Autio (1998); de la Mothe and Paquet (1998); Howells (1999); Acs (2000); Cooke et al. (2000, 2004); Asheim and Isaksen (2002); Doloreux (2002); Fornahl and Brenner (2003); Asheim and Gertler (2005); Doloreux and Parto (2005); Tödtling and Trippl (2005); Asheim and Coenen (2006); Doloreux and Revilla Diez (2007).

activities (Howells 2002; Gertler 2003). The exchange of tacit knowledge presupposes trust and personal contacts which are essentially facilitated by spatial proximity (Storper 1997; Morgan 2004). Forth, comparative studies on the governance of innovation have shown that sub-national territories differ strongly in their institutional setting and political decision making-abilities (Cooke et al. 2000).

The architecture of a RIS is of a complex nature. Based on the work of Autio (1998) we propose to grasp the structuring of a RIS by focusing on the following subsystems and crucial dimensions (see also Figure 1).

- Knowledge generation and diffusion subsystem: This subsystem comprises all those organisations that are creating and transferring technologies, knowledge and skills. Crucial actors are R&D organisations (universities, research institutes, public laboratories, etc.), educational bodies (universities, technical colleges, vocational training organisations, etc.), and technology mediating and other innovation supporting organisations (technology licensing offices, science parks, incubators, technology centers, etc.).
- Knowledge application and exploitation subsystem: Key agents in this subsystem are the
 industrial and service companies as well as their clients, suppliers, competitors and cooperation partners at the regional level. Such constellations are usually referred to as
 regional clusters.
- Policy subsystem: Government organisations and regional development agencies at the sub-national policy level constitute another RIS subsystem, providing finance and subsidies, and designing and implementing innovation and cluster policies (Cooke et al., 2000; Asheim et al., 2003; Tödtling and Trippl, 2005).
- Local flows of knowledge and skills: Ideally, there are different types of linkages within and between the RIS subsystems, leading to regional collective learning and systemic innovation. Keeble (2000) distinguishes between three key mechanisms of regional collective learning, including new firm spin-offs, labour mobility and networks. A more differentiated typology of linkages comprises market links, formal collaborations, informal networking (milieu) and spillovers (Tödtling et al., 2006). The precise nature of localised knowledge interactions, however, remains somehow disputed (Gertler and Levitte 2005; Gertler and Wolfe 2005; Malmberg and Maskell 2002, 2006; Porter 1998; Tödtling et al. 2006, Tödtling and Trippl 2007, Trippl and Tödtling 2007a).

Socio-institutional factors: The common habits, routines, practices and rules prevailing in
an area largely influence its innovation capacity, as they regulate the interactions between
the innovation actors (Johnson, 1992; Gertler, 2004; Edquist, 2005). Consequently,
institutional factors such as the dominating patterns of behaviour, the culture of cooperation or also attitudes towards innovation and technological progress are important
RIS elements.

Regional innovation systems are embedded in *national* and *international innovation systems*, which taken together influence and shape the innovation activities of firms and their abilities to absorb and exploit new technologies such as ICT (see Figure 1). RIS are core entities in the globalising economy but other "layers" of innovation also matter crucially, giving rise to a complex, multi-level architecture of the set-up of knowledge production and application. The linkages between the character of RIS and the larger institutional frameworks, however, remain little understood (Asheim and Coenen 2006). More theoretical and empirical research is necessary to examine the *impact of national institutional framework conditions* as described by the varieties of capitalism approach (Soskice 1999), the theory on business systems (Whitley 1999) and the NIS literature (Lundvall 1992, 2007) on the form and functioning of RIS. Furthermore, there is the challenge to integrate more strongly the *international dimension* in studies of RIS and to analyse the *impact of ICT on the linkages between the regional, national and global innovation frameworks*.

RIS are inserted into a complex web of relations to national and international organisations and innovation systems. It is meaningful to draw a distinction between two relevant dimensions in this respect (Tödtling and Trippl 2005): The first dimension refers to the *inflow of international knowledge and expertise*, brought about by the extra-local contacts of regional firms and knowledge providers (Bunnel and Coe 2001; Oinas and Malecki 2002; Amin and Cohendet 2004; Maskell et al. 2004, 2006). The second dimension is related to *political governance* and its multi-level character. Policy interventions and actions undertaken at the national and European levels can constitute important *external impulses*, influencing the development and dynamics of a RIS (Cooke et al. 2000; Asheim et al. 2003). The last aspect

With respect to the *distribution of competencies* between the regional, national, and European level enormous differences (with varying degrees of political autonomy for regions) within Europe have been detected (see Cooke et al. 2000). Nevertheless a pattern can be found indicating a complex division of labour (Cooke et al. 2000): At the *regional level* we can often identify competencies for the lower and medium levels of education, incubation and innovation centres, transfer agencies and, more recently, cluster policies (Boekholt

dealt with above, i.e. the multi-level-governance dimension of innovation, deserves further attention. The past years have witnessed the rise of a large number of regional, national and European policy initiatives to promote both the production and use of ICT, calling for a sound coordination of different policy levels.

Regional Innovation System Knowledge generation & diffusion subsystem NIS International organizations organizations Technology Public Educational research mediating organizations organizations organizations, Socioincubators, etc institut ional factors R&D competencies Skilled labour Knowledge application & exploitation subsystem Clusters NIS policy European Customers Contractors instruments Union policy instruments Industrial companie: Collaborators Competitors Socio-Finance - Subsidies - Innovation institut ional and Cluster Policies factors Other RIS Other RIS Regional development Policy institutions agencies Regional policy subsystem

Figure 1: Structuring of Regional Innovation Systems

Source: Own modification of Autio (1998)

5.2 RIS and ICT

ICT and the Transformation of Knowledge Linkages

There is an increasing awareness of the powerful role of ICT as an *instrument of knowledge* generation and transmission. This is related to a pressure towards the codification of knowledge and the use of computerized knowledge management systems. In particular, the Internet and "search engine" such as Google have become a widely used source of relevant information also in science and research. Furthermore, ICT can potentially be regarded to

and Thuriaux 1999). At the *national level* in many cases we find competencies for universities, specialised research organisations, and funding for R&D and innovation (OECD 1999). At the *European level* there are the structural funds, the RIS/RITTS programme, and the framework programmes for R&D and technological development (Landabaso and Mouton 2003; Oughton et al. 2002).

essentially widen the spatial scope of innovation networks. An interesting compilation of potential effects of the Internet on the innovation process of firms (see Table 1) has been provided by Kaufmann et al. (2003).

Table 1: Potential effects of the Internet on the innovation process of firms

	Increase in the efficiency of the innovation process	Change in the innovation process or extension of the innovation network		
Distribution of information	 Cheaper, faster and simultaneous distribution of information about innovation activities and within a cooperative innovation project Transmission of data which can be directly processed by the innovation partner 	 Getting into contact with new types of innovation partners Getting into contact with more distantly located innovation partners (reaching new spatial levels) 		
Collection of information	 Faster, more frequent, continuous and cheaper collection of innovation-related information Direct processing of electronic data, easier integration in in-house knowledge management Integration of internal and external knowledge systems or databases 	 New sources of information previously not aware of or not accessible due to distant location New sources of information previously not aware or not accessible due to 'relational distance' (different 'community of practice') Easier use of external databases and computational resources 		
Interactive communication	 Reduction of cost of communication in co-operative innovation projects involving distant partners More frequent and faster communication between distant partners 	 Better integration of information flows improving the knowledge management of a firm 		

Source: Kaufmann et al. (2003)

There is work suggesting that the impact of the Internet on innovation varies from sector to sector. Anderson (2001), for example, found positive effects of the Internet on innovation only in the case of dynamic and complex industries (like electronics and instruments), but not in the case of mature and low-tech sectors (such as food, clothing and furniture). Kaufmann et al. (2003) in their study on Austrian firms, however, found no support for the view that specific sectors like high-tech or producer services are able to benefit more from using the Internet in their innovation process than other firms. Another key point, raised by Kaufmann et al. (2003) is that generally the Internet is more effective for improving the communication within existing innovation networks than for finding new knowledge sources and innovation partners. Moreover, they showed empirically that the effects of the Internet concerning the spatial extension of their innovation and knowledge linkages could be primarily found at the

national and the European level, less at the global. Overall, there seems to be a weak globalizing effect of the Internet in most phases of the innovation process. *Spatial proximity* continues to matter, and both the local and the global level have their relevance as space for *knowledge interactions*. In the meantime this insight is well established in the literature. Many authors argue that both extensive relations within local clusters and RIS and strong connections to national and global knowledge sources are of importance (Bathelt et al. 2004; Gertler and Levitte 2005; Tödtling and Trippl 2007, Trippl and Tödtling 2007). This view clearly challenges the assumption of the dominance of one spatial level over another. On the contrary, Bathelt et al. (2004) have pointed out that "global pipelines" should be regarded as important complements to the "local buzz" produced in regional arenas.

RIS and the production of ICT

There are strong reasons to assume that regions and innovation systems differ regarding their ability to (1) "seed" ICT producing industries and to (2) adopt and use ICT for beneficial outcomes. Interestingly, much of the literature has been concerned with the former issue, i.e. the location and development of ICT clusters, whilst the issue of effective use of ICT has received less attention so far.

Much research has been carried out on the *geography of ICT production*, revealing a strong tendency of this sector towards a spatial concentration in clusters (Saxenian 1994; Swann et al. 1998; Keeble and Wilkinson 2000, Quah 2001; Koski et al. 2002; Acconcia and Del Monte 2003). The propensity to geographical clustering is regarded to be a typical feature of knowledge based or high technology industries (Cooke 2002).

Some authors have argued that the rise of ICT clusters is strongly related to production and transaction cost advantages (see, for example, Scott 1988). Other scholars such as Saxenian (1994) and Cooke (2002), in contrast, pointed to the importance of advantages in terms of knowledge exchange and spillovers. According to van Winden et al. (2004) the development of ICT clusters depend upon access to the benefits of a certain location, the costs of that location as well as on the role of regional and national policies (see Figure 2).

The *spatial organisation of the ICT industry* has changed considerably in the past years. The traditional centres of ICT production, which are mainly found in highly developed countries and regions such as Silicon Valley, Route 128, and Texas in the U.S., Cambridge and the South-East region in the U.K., Munich and Cologne in Germany or Paris and Grenoble in

France are facing *increasing competition* from newly emerging ICT clusters in Taiwan (Hsinchu Science Park), India (Bangalore), Korea, Hongkong and China (Chen et al. 2006; Saxenian 2005). Research has shown that the *mobility of talent* and specialists has been a key factor for the development of these new dynamic ICT agglomerations in formerly peripheral regions. More precisely, the rise of new locations of ICT production in Asia has been considerably accelerated by highly-skilled engineers and venture capitalists, who retuned to their home countries after having studied and worked abroad (Saxenian 2002, 2005, 2006). By working or creating new companies in (and, thus transferring technological entrepreneurship and first-hand knowledge of financial institutions of the new economy to) their home countries, this talent played a key role as "knowledge spillover agents" (Trippl and Maier 2007) and impelled the emergence of a new global landscape of ICT production.

ICT clusters differ in terms of their origins, development paths and structuring (Matuschewski 2006). There exists a strong diversity of ICT clusters and their development is highly context dependent and context specific. Winden et al. (2004) have provided an interesting typology, differentiating between *clusters specialised in R&D*, *local-demand-based ICT clusters* and *cost-based clusters*.

National Costs of Benefits of policies location locations Attractiveness of the region for professionals Access to generic qualified labour Transportation Access to noncosts local markets Size and Access to markets Costs of real character of the for ICT products estate/land office ICT cluster development local economy space Co-operative Transport climate Access to new connections ideas as source Taxes and Local social for "new capital benefits combinations" Quality of life Knowledge transfer Opportunities for General wage infrastructure intra-sectoral level knowledge Number of ICT spillovers firms in the city Wages for ICT professionals Access to Technical specialized universities facilities Attractiveness of Access to ICT the region for Local policies professionals ICT professionals

Figure 2: Determinants of ICT cluster development: a frame of analysis

In line with Cooke (2004) it can be argued that there are strong differences between RIS regarding their capacity to develop dynamic ICT clusters. Cooke (2004) has introduced a differentiation between traditional innovation systems (which he also calls *institutional regional innovation system* – IRIS) and new economy systems (which he refers to as *entrepreneurial regional innovation systems* – ERIS). Whilst an IRIS is well suited to promote the development of more traditional sectors with a synthetic knowledge base, high technology industries such as ICT which draw primarily from an analytical knowledge base best flourish in ERIS. The dynamism of ERIS rests – in sharp contrast to IRIS – on local venture capital, entrepreneurship, scientific excellence, market demand and incubators which support intense processes of knowledge exploitation.

Innovation processes in the ICT industry exhibit specific features, differing strongly from those in more traditional sectors as regards key knowledge sources, the role of codified and tacit knowledge and the types of knowledge links and local clustering (Asheim and Gertler 2005, Tödtling et al. 2006). Like other knowledge based sectors such as biotechnology, the ICT industry is regarded to be dominated by an analytical knowledge base. There is a strong

reliance on scientific inputs and codified (or codifiable) knowledge is in general far more important than in traditional sectors which rely on a synthetic knowledge base (for an overview about the main features of analytical and synthetic knowledge bases see Table 2). An analytic knowledge base also implies that knowledge inputs are often derived from reviews of existing (codified) studies, knowledge generation is based on the application of widely shared and understood scientific principles and methods, knowledge processes are more formally organised (e.g. in R&D departments) and outcomes tend to be documented in reports, electronic files or patent descriptions. Although the codification of knowledge plays a decisive role in sectors with an analytical knowledge base, tacit knowledge is of relevance, too. In ICT and other knowledge based sectors there is much more systematic basic and applied research than in traditional industries. The rate of product and process innovations, notably of a radical nature, is high. R&D efforts are typically focused on generating radical innovations. Academic spin-offs and new firm formation are important mechanisms when it comes to the application and economic exploitation of new analytical knowledge. Research is done to a considerable extent within companies. Nevertheless innovating companies are highly dependent on external knowledge sources. Universities, government labs and other research institutions are crucial agents in this respect, providing scientific research inputs for innovating firms. Consequently, various forms of university-industry partnerships play a pivotal role in the process of knowledge generation and innovation.

Table 2: Synthetic and analytical knowledge bases

Key features						
Synthetic knowledge base	Analytical knowledge base Knowledge-based industries (e.g. biotechnology, ICT)					
traditional industries (e.g. industrial machinery, engineering)						
Dominance of tacit knowledge and practical skills	 Dominance of codified (codifiable) knowledge complementary role of tacit knowledge 					
Application or novel combination of existing knowledge	 Application of widely shared and understood scientific principles and methods 					
Low levels of R&D	 Systematic basic and applied research, formall organised knowledge processes (e.g. in R&D departments) 					
Strong orientation on solving specific problems articulated by customers	 Strong reliance on scientific research inputs from universities, government labs and other research institutions 					
 Learning by doing and interacting, user- producer relationships 	 Learning by exploring, university-industry partnerships 					
Incremental innovations	Radical innovations					

Source: Tödtling et al. (2006)

Recently, Trippl and Tödtling (2007b) have also argued that regions differ strongly in their capacity to promote the development of high technology sectors such as ICT. They propose a theoretical framework that links the rise of clusters to a dynamic analysis of regional innovation systems. Departing from the theoretical concept of regional innovation systems (RIS), a distinction between "RIS with strong potentials for high technology industries" and "RIS with weak potentials for high technology industries" is drawn. The key thesis suggested by Trippl and Tödtling (2007a) is that the development pattern of regional high tech complexes is strongly dependent on the strengths and the structuring of the respective RIS. Regions that already host successful high technology industries constitute a favourable environment also for the rise of new knowledge intensive clusters, even if the newly emerging sectors are different from those developed in the past. These areas are well endowed with generic factors such as excellent universities, knowledge mediating institutions, venture capital organisations and highly skilled mobile labour. Other key features of such regions often include a culture of academic entrepreneurship and high risk taking, a propensity to cooperate and share knowledge and positive attitudes towards innovation and technological progress. In such "RIS with strong potentials for high technology industries" the emergence and growth of a high technology cluster might be a spontaneous phenomenon, as it could build on existing generic functions and expertise necessary for "seeding" high technology sectors. Due to the tradition of these areas as high technology centres, a considerable body of knowledge is available at the local scale. Consequently, it can be suggested that knowledge residing within the region and its local circulation play a crucial role. The case of California is telling in this respect. Prevezer (2001, p. 18) analyses the emergence of the biotechnology sector in this region and shows that the industry "inherited a great deal from the earlier development of computing" in the area. Several of the prominent preconditions for successfully developing high technology companies were there, including excellent research organisations, experienced venture capitalists, a pool of highly skilled mobile labour, and good communication networks. Consequently, "the history of having grown the computing industry in California was relevant to the establishment of biotechnology in the Bay area" (Prevezer 2001, p. 25). Boston with its transitions from electronics, to computers and software, to biotechnology and where generic elements (research universities, venture capital, networks) have supported this transformation is another prime example for a "strong high technology RIS" (Tödtling, 1994).

In regions which have no tradition in promoting high technology industries and which have to be regarded as latecomer in a specific technological field, the rise of knowledge based sectors such as ICT is likely to take a different route. The argument is not in favour of building ICT clusters from scratch. Instead, the focus is on regions which possess important factors such as excellence in science or other competences rooted in the area on which a cluster can grow but lack the critical mass for spontaneous take-off. Enright (2003) refers to such constellations as "potential clusters". Due to the regions' weak commercialisation capabilities, they fail to capitalise on the existing assets. These areas often have little experience in commercialising scientific discoveries, a weak culture of risk taking, low levels of social capital, and frequently they lack crucial factors such as venture capital or a support structure specialised in promoting academic spin-offs. Tödtling and Trippl (2007a) refer to such settings as "RIS with weak potentials for high technology industries". The key thesis proposed is that in such regions, the RIS must undergo a far reaching transformation for high technology clusters to emerge. Such RIS changes become manifest in the creation of a variety of new organisations, processes of institutional (un)learning and socio-cultural shifts. There are good reasons to assume that the state plays a stronger role in such regional settings to promote knowledge based clusters.

The rise and growth of knowledge based sectors in "RIS with weak potentials for high technology industries" is strongly linked to the capacity of the region to "rebuild" its RIS. The

transformation of a regional innovation system is a complex and multifaceted process. The following two key dimensions deserve further discussion: Firstly, an adaptation and renewal of the institutional set up brought about by "implantation" of new elements is crucial in this context. More specifically, to promote excellent research and education institutes and to establish supporting agents such as science parks, academic spin-off centres, or technology licensing offices that are specialised in encouraging start-up companies are important preconditions for the development of the ICT industry. To strengthen the ensemble of specialised organisations, however, covers only one aspect of the reconstruction of a RIS. New routines, attitudes and patterns of behaviour must also emerge. To initiate changes in these "soft" institutions might be the most challenging endeavour in the process of creating a favourable environment for high technology industries. It is of special relevance in the knowledge generation and diffusion subsystem, reflecting the need that researchers learn to commercialise their scientific work by adopting more positive attitudes towards new firm formation and cooperation with industry. In the policy system also new routines seem to be of utmost importance. To encourage high technology industries, one cannot rely upon old policy recipes and traditional instruments such as subsidies, tax incentives or low cost labour (Audretsch, 2003; Feldman and Francis, 2004; Florida, 2005). It requires a substantial amount of policy learning, leading to a new mode of state engagement that is about investments in the knowledge infrastructure, and establishing conditions that attract talent and secure the availability of venture capital. Other measures to encourage entrepreneurship and to stimulate innovation interactions to promote a steady flow of knowledge at the regional scale may also deserve attention.

RIS and the use of ICT

ICT is regarded to be a generic technology, affecting *directly* and *indirectly* the entire economy and society. Looking at the level of firms, Bocquet et al. (2007) provide empirical evidence suggesting that the adoption of ICT is strongly related to firms' strategies, to their organizational practices and to their competitive environment.

As already noted by Freeman and Perez (1988) the diffusion of basic technologies such as ICT, is inextricably linked to and critically dependent on far-reaching *social and institutional changes*.

There are, indeed, strong reasons to assume that the provision of hard forms of infrastructure is far from being sufficient. In the past decade it has become clear that a simple "transplantation" of ICT onto regional economies is not a viable strategy. The impact of ICT on growth is strongly dependent on the socio-economic conditions prevailing in the respective region. Key factors in this context include, for example, the production structure, i.e. the size of local companies and their embeddedness in larger supplier/customer networks. It is important to note that advanced networks cannot be formed by creating an ICT infrastructure if no networks existed before, pointing to the limits of a supply side approach. Beyond hard infrastructure also organisational quality and learning and innovation attitudes in companies, supporting organisations and in the policy system are vital, i.e. the existence of "soft infrastructure" elements.

It can be suggested that RIS differ enormously in their capacity to realize processes of institutional (un)learning and to transform themselves to become favourable environments for the adoption of ICT. Indeed, there is evidence that, while the potentials of ICT are in principle available to every area, the ways and the effectiveness with which regions exploit these potentials differ enormously across Europe (Cornford et al. 2006).

According to the i2010 High Level Expert Group (2006) there are several factors which have a strong bearing on the effective use of ICT. These include:

- appropriate skills,
- organisational change,
- scope for experimentation,
- appropriate management practices.

Cornford et al. (2006) assume that *regional innovation cultures* are a central key to unlock the potential of ICT in regions. More specifically, the authors point to the importance of various key factors, influencing the capabilities of regions to use ICT in an effective way. The most relevant factors or dimensions identified by Cornford et al. (2006) comprise:

- the proliferation of loosely articulated networks (open network structure) that enable the construction and propagation of meanings and values,
- individual, institutional and collective learning,

-

We are grateful to an anonymous referee for the following arguments on the importance of soft infrastructure elements.

- the development of shared representations of possible futures, collective vision and reflexivity,
- leadership,
- a sound combination of regional openness and closure to the outside world.

Other RIS elements such as strong ICT research capacities, a well-functioning ensemble of educational organisations and knowledge transfer institutions specialised in ICT, a highly-developed local ICT producing industry, the establishment of local and global linkages for ICT diffusion as well as public authorities actively promoting the adoption of ICT might determine the "absorption capacity" of regions to exploit ICT in effective ways. The role of each of the factors listed above and their interplay, however, remain poorly understood and need both further theoretical efforts and empirical investigation.

The development of ICT also feeds back onto *the functioning of innovation systems* and RIS in particular. Many central innovations in ICT were triggered by the specific needs of components of innovation systems, by universities and researchers. So were the first Internet connections established between four US universities in order to allow them to cooperate more directly on defence projects. Later on the US National Science Foundation invested research money into a backbone network infrastructure and supported universities who wanted to link up to this network under the condition that "... the connection must be made available to ALL qualified users on campus." (Leiner et al., 2003) With this condition NSF for the first time opened up the network technology to a broader set of potential users.

At these times, the exchange of email messages, discussion in netnews and file transfer via FTP were the key applications. The respective software was closer to the needs of the machines than those of the users and therefore difficult to use. But even at these times, it was the needs and ideas of the research community which led to new proposals, new services and new, usually more user friendly, software.

A major breakthrough occurred in 1990, when Tim Berners-Lee circulated a proposal for an Internet based information management infrastructure which initiated the development of the World-Wide-Web (W3C, 2000). This Internet service became so popular within a short period of time that nowadays many people erroneously view the world wide web as the Internet. This development on the one hand triggered myriads of commercial services and on the other hand also provided the infrastructure for the development of the open source

movement and triggered many contributions resulting from hours of voluntary work by "the community".

Today, the functioning of a RIS without the support from ICT in general and the Internet in particular is almost unimaginable. We can distinguish four major tasks in the functioning of a RIS that are strongly supported by ICT use:

- 1. information,
- 2. communication and coordination,
- 3. access to resources.
- 4. cooperation and collaboration.

As far as *information* is concerned, practically all members of a RIS in the developed countries provide extensive information on the Internet. This applies to universities and firms as well as to policy institutions and specific policy initiatives. It goes without saying that the use of ICT by the public policy system is not confined to the provision of information but it is far more multi-faceted as the discussion on the rise of different forms of e-Governance has shown.¹⁵ The information is stored on servers and controlled by the respective institution. It typically serves various purposes. In addition to informing the general public such webpresences are used as instruments for marketing, public relations, image building, etc. Most of the time such information infrastructure is self contained and includes few or even no links to information provided by other institutions. The opportunities for integrating information provided by others are utilized only to a limited extent. In this respect non-commercial, open source oriented information providers tend to be more open than others.

Communication and coordination are essential elements of a well functioning RIS. ICT provides many opportunities in this respect. They range from email messages and text messages sent to mobile phones to newsgroups, database-based membership systems and other forms of electronic communities. The electronic form of the communications allows the use of programs for routine tasks.

Access to resources mainly refers to knowledge resources. The information provided on the web grows continuously and constitutes a valuable resource for others. Of course, only codified knowledge can be provided in this way, which implicitly increases the relative importance of tacit knowledge.

We are grateful to an anonymous referee for this comment.

While originally the quality of the information on the Internet was difficult to assess, in recent years more and more mechanisms of quality control have been developed and implemented. Most of them follow the principle of peer judgement and ask users to judge the quality of the respective piece of information. The published judgement then serves as a guideline for new users. These mechanisms point in the direction of more interactive forms of web use. Such technologies are often referred to as "web 2.0". A boost in quality also resulted from the increasing engagement of established media providers in electronic media. Most relevant scientific journals nowadays offer electronic access to their stock of peer-reviewed journal articles, in some cases clearly shifting emphasis away from the print version towards the electronic one.

Key players in electronic access to resources are the *indexing databases*. Services like Google have developed sophisticated indexing methods for all kinds of electronically available information and typically serve as entrance gates for Internet searches. From the point of view of a RIS, electronic access to resources serves as the link to the outside world that allows the integration of – codified and electronically available – knowledge from all over the world. To put it differently: The emergence of advanced ICT allows for the transfer of information and codified knowledge over long distances and, thus, enables regional actors to get access to ideas, knowledge and expertise that are not generated within the limited context of the regional innovation system. The circulation of tacit knowledge, which is acknowledged to be vital in the innovation process, is, however, still critically dependent on face-to-face contacts and spatial proximity.

The more recent developments in Internet services (Web 2.0) typically support electronic cooperation and collaboration among users. In Blogs, Wikis and similar services, the traditional differentiation between information provider and information user on the web becomes more and more blurred. The interactive services allow information users to become actively involved in the creation and servicing of information. They can evaluate, comment and often even alter the information provided by others. These services are currently developing very rapidly both in technical terms and in terms of standards and norms among users.

As this discussion shows, ICT provides many opportunities for improved functioning of a RIS. Most ICT based solutions are dramatically cheaper than corresponding non-electronic – mostly paper-based – versions. ICT makes it easier for a RIS to link up to the global pool of

knowledge and expertise. The availability of information, however, does not necessarily mean that it can directly be used in the RIS context. A major task of the institutions of the RIS is therefore, to translate and adjust this information for the RIS context. Moreover, one should also not forget that only a specific type of knowledge – codified and electronically available – can be accessed in this way.

6 Globalisation and ICT in the Knowledge-based Economy

ICT is a key factor driving economic globalisation and associated economic changes. Innovation in ICT is a critical component which can reduce both fixed and variable transactions costs of market entry and, more generally, economic performance. However, while the remarkable changes in the capacities of machines to process, communicate and store information have led to enormous efficiency and other gains, the potential downside of such rapid changes has been very rapid rates of physical and human capital depreciation. Accordingly, in such rapidly changing environments, there can be a high premium to having appropriately qualified human capital, which embodies the most recent technologies. In general, it is critical for the adaptability of a country or region that there be a continuous upgrading of human capital formation, technological readiness, and technological infrastructure investments. New technologies provide an opportunity for agents and regions to "leap frog" the shortcomings of their existing technologies and close performance gaps relative to that of technological leaders. However, existing technological positions may be reflecting underlying human capacity, financial and other constraints, which can not be reversed without sustaining investment policy initiatives over the medium and long-term. Accordingly, in the shorter term, technological change may be associated with significant hysteresis effects, whereby present levels of performance are essential for understanding a path dependency in rates of ICT diffusion. It should be noted that the size of firms may be a critical determinant of their access to resources, which are essential to investing and risk taking in order to promote ICT adoption.

Globalisation entails both increased international strategic interdependence and performance. It is a process through which a growing number of economic agents operate in an increasingly worldwide market place. As a consequence more and more agents find that their economic decisions have to take account of not just the increased opportunities, but also the increased competition, arising from an expanded range of relevant international locations that spans the planet. Thus, globalisation is inherently a double-edged sword. On the one hand, there are potentially increased opportunities, notably for heightened competitiveness and increased market sales over an expanded international geographic space. For example, relocation of different segments of a firm's activities can facilitate access to lower cost production sites and unique human capital and other resources, as well as lower cost

intermediate goods and service suppliers. On the other hand, however, there is also increased pressure for local market restructuring in order to gain competitiveness. In that regard, labour markets often constitute an Achilles' heel of the economic adjustment processes triggered by globalisation. Not only can labour, typically, be much less mobile than capital, or many types of goods and services, but also it is often difficult, or even impossible, to substantially upgrade human capital skills in the short to medium term. Thus, it should be emphasized that globalisation often entails both "winners" and, at least some, "losers". In the ICT context, this is clearly illustrated by the policy predicament of technologically disadvantaged countries and/or regions facing the "digital divide".

A consideration of some of the principal determinants of globalisation can offer valuable insights regarding its consequences and the design of optimal policy responses. Notably, to admittedly different degrees, many countries and regions have shown heightened willingness to embrace market liberalization, openness and facilitated market access. In general, reduced transactions costs hampering domestic and international trade in goods and services is an essential contributing factor to both the globalisation process and the relative competitiveness of specific countries and regions. In certain instances, such reduced costs have been accomplished through the dismantlement of various forms of government intervention and regulations which have previously hindered the free flow of goods and services. Procompetitive policies have typically included the lowering of tariffs and discriminatory tax rates, the reduction or elimination of such non-tariff barriers as government standards, procurement policies and subsidies. Other factors which can reduce trade costs are pro-active government policies promoting market access, as through reduced transportation costs. Other dimensions of effective governance, such as excessive regulatory burdens and legal frameworks, can also be crucial for fostering a vibrant and internationally competitive business environment. Critically, in an increasingly linked and competitive international economic environment, there can be high costs to policy inertia and the maintenance of segmented market conditions or other facets of the status quo, which are anti-competitive.

In many European countries the perceived social and governance constraints, associated with the functioning of labour markets appear to have generated more acute policy tradeoffs. The relative rigidity of certain EU labour markets can be contrasted with those in US and many Asian and/or developing countries.

6.1 ICT-enabled Outsourcing, Offshoring, and Firm Performance

The ICT is a major driving force for outsourcing and offshoring. The rapid advancement of technology, in particular ICT, and the worldwide deregulation and competition in the telecommunications industry have led to a substantial decrease in adjustment and communication costs faced by firms. Firms no longer can be viewed as single entities that produce final goods. Increased global connectivity and the resulting international cost saving opportunities for production mean that firms now face decisions to outsource and offshore components of their production process. Thus, Grossman and Rossi-Hansberg (2006) argue that there is a need for a new paradigm in international trade theory that places 'task' trade at the centre.

Many researchers have studied the implications of *increased openness and reduced costs of outsourcing* on the vertical integration decisions of firms. The literature on the "fragmentation" of the production process has provided interesting insights into the effects on trade flows, welfare changes and factor prices. Another strand of this literature examines the *formation of international hierarchical production teams*.

Lower communication costs are associated with higher levels of international offshoring. The model on the formation of hierarchical production teams proposed by Antras et al (2006) highlights the important effect of communication technologies on the characteristics of international offshoring. However, the quality of offshoring is lower. Furthermore, the model provides insight into how globalisation, viewed in terms of the formation of cross-country teams, affects the organisation of work, size distribution of the firms, the structure of earnings of individual. They show how these outcomes impact on production, consumption and international trade in the global economy. The less skilled agents specialize in production and more skilled agents specialize in problem solving. Globalisation leads to better matches for all southern workers but only for the best northern workers. As a result, globalisation increases wage inequality among non-managers in the South, but not necessarily in the North.

-

Outsourcing is an arrangement in which an outside company provides activities for a company that could be or usually have been provided in-house. Offshoring is a subcategory of outsourcing. Offshoring refers to outsourced activities that are conducted abroad. Further classifications of outsourcing based on location and control/ ownership criteria include; captive onshore/ non-captive onshore outsourcing which refers to a shift from intra-firm supplies to an affiliated / non affiliated firm in the home economy. Captive / non captive offshoring refers to sourcing of activity from an affiliated / non affiliated firm abroad (World Trade Report, 2005). Metters et al (2007) discusses the current state of affairs in offshoring and the factors that have created the present environment. They argue that US government neglect, foreign government activism, technological change, cultural change in relation to services processes among business people and cultural relationships among countries combined to create the current environment of services offshoring.

It has been argued that European countries slow take up of ICT is acting as a barrier to fragmentation and specialisation of business processes. Abramovsky and Griffith (2005) examine the role of ICT investment in influencing a firm's decision to outsource and offshore services and find that ICT intensive firms purchase greater amounts of business services on the market. They also find that these firms are more likely to purchase business services offshore. Specifically they find that both ICT investment and Internet use increases the probability of a firm offshoring by 12 %.

Bhalla et al (2007) investigate *the link between a company's performance and the extent of its offshoring of IT-enabled services*. However, they fail to find such a link and argue that further research is warranted into what are the expected benefits from offshoring and when is the optimal time for a firm to offshore. Gorg and Hanley (2004) examine the relationship between *outsourcing and profitability at the level of the plant*. They find that on average, other things equal, larger plants in the manufacturing sector benefit from outsourcing materials inputs while small plants do not. Results for the service sector are not clearcut.

A large proportion of service sector occupations are found to be potentially offshorable. Van Welsum and Vickery (2005), apply the following classification of occupations based on four "offshorability attributes" (i) intensive use of ICT (iii) an output that can be traded and transmitted in a way that is enabled by ICT (iii) level of codifiable knowledge and (iv) no face-to-face requirements. Based on data for several OECD countries they calculate that 20% of total employment carried out functions that could potentially be offshored as a result of rapid technological advances and increase tradability of services.

There is a large variation in the estimates, provided by other studies which have conducted similar analyses, of *total employment potentially offshorable*¹⁸ Bardhan and Kroll (2005) classify employment based on job characteristics and estimate that 11 % total employment in the US in 2001 was offshorable compared to a figure of 44 % estimated by Forrester Research and reported in Kirkegaard (2004). The large variation is mainly due to differences in occupation classification criteria. Jensen and Kletzer (2005) identify potentially tradable service sector occupations based on spatial clusters and estimate that 30 % of employment in the US may be affected by offshoring. They fail to find conclusive evidence of weaker employment growth in tradable occupations than in non-tradable activities.

_

¹⁸ Summarised in van Welsum and Vickery, 2006, Mankiw and Swagel, 2005.

From a policy perspective, it is interesting to uncover the *underlying factors of the proportion* of offshorable occupations in total employment. Van Welsum and Reif (2006) use data from 12 OECD countries (including 9 EU countries) over the period 1996-2003 and estimate the factors driving the shares of potentially offshorable clerical and non-clerical occupations in total employment. They distinguish between FDI in manufacturing and services. 19 The grouping of occupations is very relevant as the clerical group include jobs that can be substituted for by ICT therefore a differential pace of adoption and integration of technology can have a different effect across countries. The results suggest that the share of exports of business service in GDP, the share of ICT investment in total gross fixed investment, share of services sector in GDP and human capital are positively associated with an increase in the share of employment in potentially offshorable non-clerical occupations, while the share of imports has an offsetting effect on the share of employment in potentially offshorable nonclerical occupations. They find that an increasing share of employment in potentially offshorable clerical occupations, the exports to GDP ratio, the human capital measure, share of hi-tech output in GDP, and product market regulations are positively related. Imports of business services, declining trade union densities and rising share of services in GDP are negatively related.

Bunyaratavej et al. (2007) investigate the determinants of the location of services offshoring. They relate services offshoring to the literature on international business research of the cost of doing business abroad (CODBA), liability of foreignness (LOF) and institutional theory. They find that important location factors for services outsourcing are lower labour costs and human capital while due to telecommunications technology proximity to major markets is less vital. Institutional theory emphasises the critical role *institutions* play in an economy lowering transaction costs and information costs and facilitating interactions. Locating in countries with similar culture, political systems, economic systems, legal systems to the home country should reduce CODBA/LOF and make easier the firms' integration with the local institutional environment. Bunyaratavej et al. (2007) find that firms are more likely to offshore to locations where wages, culture, education and infrastructure closely resemble their home country.

This paper is an extension of their previous analysis van Welsum and Reif (2006)

6.2 The Impact of Globalisation and ICT on Regional Economic Performance

Globalisation both reflects and impacts a number of dimensions of firms' and other entities' domestic and international economic performance. From the perspective of regional economic performance, globalisation potentially not only offers unique opportunities for regions to reap returns from unique assets on expanded, increasingly worldwide markets, but also permits them escaping the constraints of unfavourable local economic conditions. *Reduced international transaction costs*, facilitated by the use of ICT, constitute a critical component of globalisation. Such cost savings can redefine the nature and functioning of markets, their interconnectivity, as well as the relative importance of spatial economic factors. As a result, there are potentially asymmetric changes in certain of the relative costs of agents, which are often interrelated in complex ways with spatially sensitive economic factors and, in turn, agents' competitiveness. For example, while a firm in a remote region, may be able to effectively use e-marketing to sell a unique product, its ability to satisfy customers could be dependent on complementary infrastructure investments which would reduce the transportation costs for the delivery of the goods to potential clients.

The progressive build-up of *trade and FDI* are interrelated with other specific processes such as *outsourcing, fragmentation, offshoring or delocalization of economic activities*. As previously discussed, one both positive and negative aspect of globalisation is increased pressures for international competitiveness. As Michael Porter and other have emphasized, this leads firms to decentralize their value-added chains on a global level in order to avail themselves of sources of reduced costs and/or unique resources. Broadly speaking, ICT can be viewed as countering traditional spatial constraints, which required the bundling of complementary activities in specific locations. ICT facilitates the locational dispersion, fragmentation and decentralization by reducing organizational costs of coordination. This applies to the exchange of goods and services both within and outside institutions' original structures.

In the case of *outsourcing*, for example, ICT can permit the substitution of potentially lower cost external suppliers which can have cost advantages relative to internal supply by an institution envisaging such outsourcing. Traditional reasons for such external cost advantages include access to unique technologies, resources, including skilled and unskilled labour, as well as suppliers' ability to achieve economies of scale and/or scope by serving multiple

clients. Specifically, outsourcing is often driven, to a significant extent, by low efficiency wages, as illustrated by the case of software consulting services provided in Ireland and India. These in turn depend on the supply of well-qualified workers, who can be both locally and foreign trained and/or recruited.

ICT can be viewed as vital to the effective communication, processing, and use of information permitting the *efficient coordination* of the timing of external supply, relative to an institution's internal demands. Moreover, there are often quality control issues, which can require the sharing of potentially complex information regarding the nature of production processes. An essential insight here is that *globalisation entails a real challenge to the measurement of ICT's contribution to an institution's competitiveness*. Specifically, an assessment of the overall contribution of ICT investment to enhanced productivity and profitability may be obfuscated by the fact that ICT is interrelated with many different dimensions of the institution's activities and structure.

6.3 Policy Stakes of Globalisation, ICT and Regional Development

The heightened economic interdependence entailed by economic globalisation provides a strong rationale for proactive government policies, notably to promote technological innovation and to foster knowledge-based economic activities. Greater policy cooperation and coordination, both between countries and regions within and between countries, is required for a variety of reasons. These include a shift in the relative heightened strength of markets, as compared to individual governments and international institutions. In large part, this is due to increasingly high degrees of capital mobility and greater competition between countries and regions on an international scale, in order to attract the many faceted activities of multinational corporations.

Economic globalisation is also associated with more *market failures*, with associated greater scope for *negative international externalities*, as well as *inter and intra-regional*, *externalities*. Such externalities can be either positive, or negative, as in the cases, respectively, of R&D spillovers or job losses. There is also an increased *vulnerability to foreign market and government failures*, as illustrated by the case of pirated computer software in many countries worldwide.

Globalisation also entails *heightened opportunity costs of poor policy choices and design*. In part, this is, as previously discussed, due to the increased pace of technological change. The

heightened interdependence linked to economic globalisation is redefining the gap between technological frontier countries/regions and economic leaders, on the one hand, and technological followers and poorer countries/regions. Over time there can be major distributive effects, potentially leading to sharp policy tradeoffs between policies favouring competitiveness and social redistribution. In addition, policy stakes are often accentuated by marked asymmetries in economic agents' initial economic positions. Thus, the *policy options* available to agents may be quite dependent on their initial wealth, as illustrated by the case of relatively poor and rich regions within and across countries. Such asymmetric positions can mean that the "core" of coordinated policy solutions at international levels may be relatively limited, but these may be easier to resolve at regional levels given heightened perceptions of shared communities of interest. Thus, proactive interregional and intraregional policies may be first-best policy responses. Longer-term economics stakes can be defined by potentially virtuous or vicious economic growth paths.

Thus, it appears that *proactive country and regional policies favouring innovation and ICT diffusion* are increasingly of primordial importance. The development of appropriately skilled human capital can be crucial for success, as illustrated in Ireland and Singapore or by the cases of Silicon Valley, Seattle, as well as by Cambridge UK or USA. Effective policies need to support mechanisms that generate a concentration of certain economic activities in order to realize increased scale economies and/or agglomeration effects, which correspond to the internalization of positive location-specific spillover effects (i.e. externalities).

In an increasingly knowledge-based economy, worldwide, the latter include R&D spillovers, which can be linked to a variety of factors including the *quality of national and regional innovation systems*. More specifically, *technological competitiveness* can be critically impacted by university-industry linkages, the state of human capital development, labour-market synergies linked to complementary training in different companies, etc. In general, many regions tend to have more specific centres of specialized activities, so that the former strategies may be preferable. Nonetheless, the financial and other constraints, arising from smaller sizes and resources, also suggests the need for regions to foster intra and interregional networks, in order to achieve scale economies and to foster positive knowledge and other spillover effects.

7 Summary and Conclusions

7.1 Summary

The Economic Impact of ICT Diffusion

The rapid development, adoption, and use of ICT innovations have transformed economies at the national and the regional level and all their sectors. In this respect, ICT functions as a new general purpose technology, which impacts economies both broadly and deeply by generating a wide array of new products, production processes and services. Furthermore, ICT has given rise to new industries within both the manufacturing and the service sector. However, it is a common feature of new general purpose-technologies that it takes a long time before they are implemented (including the necessary organisational changes) and used in such a way that they could develop their abilities to the fullest.

The adoption of ICT allows the reduction of transaction costs and leads possibly to more efficient markets. The emergence of new goods and services as well as changes in the characteristics of old goods and services due to the use of ICT, including the ways goods and services are produced and distributed lead to changes in market structures and competitive conditions affecting and creating new opportunities for small firms and entrepreneurs. As ICT are routinely deployed in organisations to re-engineer processes, gain new strategic advantages, or network across organisational boundaries, they change both the internal organisation of companies and other organisations and the relationships between companies and organisations.

In addition, the spread of ICT is changing the labour market by generating new ICT-occupations and at the same time changing the requirements for non-ICT jobs. Due to the structural changes generated by ICT, employment is increasing in some sectors and declining in others.

There are *three main channels* through which ICT affects economic growth. The first channel is the *emergence of new sectors* embodying *new technologies* including the ICT-producing sectors themselves. In this case growth comes from two sources: (i) new sectors exhibit higher growth rates of value added, productivity and incomes and will thus function as a source of growth for the whole economy, and (ii) new sectors change the conditions of other

sectors of the economy by changing relative prices, and by providing a new set of inputs that raises productivity either by the introduction of new or improved products or new production methods. The production of ICT and the emergence of new ICT-based industries contribute directly to increase GDP and to boost aggregate productivity.

The second channel is *increased investments in ICT* by companies and governments which lead to labour and total factor productivity growth. Investments in ICT complement or replace investments in other capital goods and increase the capacity of the production of ICT-using sectors and industries.

The third channel has an indirect growth impact, namely *spillover effects*. Spillover effects of technological advances from industries producing ICT to industries using ICT result in an increase in total factor productivity.

The benefits of investing in and using ICT depend on sector-specific effects. ICT are more important to raising productivity in certain sectors than in others and since different countries have different sectoral specialisation, their gains from investing in and using ICT will differ. Investments in and use of communication network technologies present a special case because of the benefits derived from spillover and network effects. Furthermore, since the conditions to develop and to use ICT vary substantially between regions, we expect *large variations between regions* in the timing as well as in the extent that they are affected by ICT.

Regional Dynamics and Transformations during the Deployment Phase of ICT

Locational choice is increasingly becoming governed by access to particular skills, technology, and knowledge, as well as entrepreneurial talent and venture capital. Of particular importance is the provision of ICT skills, ICT technology, ICT knowledge, ICT services, ICT entrepreneurial talent, and ICT competent venture capital. This follows from the fact that in most advanced economies an ever increasing share of economic inputs and outputs is in the form of ICT and knowledge.

The increased use of ICT enables major reductions in geographical transaction costs by reducing spatial information frictions. Examining the interrelationships between three variables – increasing returns due to scale economies, demand for final products and geographical transaction costs – in a world with monopolistic competition makes it possible to draw some general analytical conclusions concerning the effects ICT-induced reductions of

geographical transaction costs. When geographical transaction costs are reduced, producers in large regions, i.e. regions with large home markets, which already have good opportunities to exploit economies of scale due to a large home market, can lower the production costs by also delivering to other regions, i.e. by increasing their exports. When exports increase, there will also be increases in incomes, which induce more producers of differentiated products to start production in the large region. Increased exports also imply an increased demand for differentiated inputs, which will induce more producers with their internal scale economies to start producing such inputs. As a consequence, we have a situation with cumulative causation or positive feed-backs initiated by the effects of ICT on geographical transaction costs. It follows that investments in ICT and particular in communications equipment stimulate further agglomeration.

The literature on innovation systems strongly indicates that knowledge flows, including spillovers are at the core of regional development. Since knowledge sources have been found to be geographically concentrated, location is crucial in understanding knowledge flows. In addition, the capacity to absorb flows of new knowledge is facilitated by geographical proximity.

Large, dense regions offer special advantages in terms of knowledge flows and knowledge spillovers, since they combine the localisation of clusters in specific industries with industrial diversity, i.e. with a range of different industrial clusters. When a (large) functional region has achieved an initial advantage in knowledge production due to e.g. a large pool of well-educated labour and a rich supply of ICT capital assets, it will attract (i) knowledge-creating and knowledge-utilising firms, since it offers opportunities to take advantage of increasing returns in knowledge production and knowledge use including imitation, and (ii) knowledge-rich labour, which wants to take advantage of the increasing demand for its skills. With increased knowledge intensity in larger regions we can expect increased investments in ICT capital assets, which will further reduce geographical transaction costs.

Regional Innovation Systems and the Layers of Innovation

ICT and regional systems of innovation are closely connected and mutually influencing one another. Regional absorption and application of ICT is an important factor for the functioning of the respective regional innovation system (RIS). At the same time, the structure of the RIS

is a major location factor for ICT industries of various types and may also stimulate the regional adoption and exploitation of specialized forms of ICT applications.

The innovation system approach views innovation as an evolutionary, non-linear and interactive endeavour that requires intensive interaction between actors. These interactions can substantially be supported by the application of various forms of ICT. The innovation system approach also highlights the importance of formal and informal institutions for the innovation process. In recent years regional scientists have emphasized that many of these interactions are spatially bounded thus leading to a strong regional component in innovation systems. Consequently, they argue in favour of regional innovation systems (RIS). Two key arguments in this respect are the spatial limitations for knowledge spillovers and the importance of tacit knowledge for many innovations.

RIS is a complex system of various interrelated subsystems: the knowledge generation and diffusion subsystem, the knowledge application and exploitation subsystem, the policy subsystem, the local flows of knowledge and skills, and the socio-economic factors. In addition to this complex structure, regional innovation systems are also embedded in national and international innovation systems and thus linked to global trends of economic development. These linkages influence and shape the innovation activities of firms and their abilities to absorb and exploit new technologies such as ICT. The chapter underlines the complex, multi-level architecture of knowledge-production and application and their dependence on national and international institutional framework conditions. Potentially, ICT plays a crucial role in linking these layers of the innovation system.

The relationship between ICT and RIS elements is analyzed from three angles: first, ICT and the transformation of knowledge linkages, second, RIS and the production of ICT, and third, RIS and the use of ICT.

As far as the first aspect is concerned, *ICT* is viewed as an important instrument of knowledge generation and transmission. The intrinsic advantages of ICT in storing, transforming, and transmitting information exerts pressures towards the codification of knowledge and the use of computerized knowledge management systems. Science and research widely use the Internet and electronic search engines as sources of information. In this way the availability of those instruments that were to a large extent developed by science and research widens the spatial scope of innovation networks. Empirical evidence, however, shows that ICT is more

effective for improving the communication within existing innovation networks than for finding new knowledge sources and innovation partners. Nevertheless, ICT helps to establish the "global pipelines" that are important complements to the "local buzz" produces in regional arenas.

The geography of ICT production, the second angle used in the chapter, has received much attention in research, revealing a strong tendency of the sector toward spatial concentration in ICT production clusters. The historical spatial organization of the ICT industry has changed considerably in recent years. New ICT clusters have emerged in Asian countries, in some cases stimulated by the mobility of specialists who returned to their home countries. Such spillover agents seem to play an important role in the creation of new ICT based clusters.

The empirical evidence about ICT production clusters reveals differences in the economic base of such clusters. Some are motivated by local R&D, some by local demand, and some by low cost in the area. Also, regional innovation systems differ in their ability to generate ICT clusters. While traditional regional innovation systems seem to be better suited to promote more traditional sectors because of their reliance on a synthetic knowledge base, R&D driven ICT production clusters require an analytical knowledge base that is argued to flourish best in a so called "entrepreneurial innovation system" which emphasizes local venture capital, entrepreneurship, scientific knowledge, market demand, and incubator institutions. The ICT industry is regarded to be dominated by an analytical knowledge base, relying on scientific inputs and codified knowledge. R&D efforts in a RIS targeted toward ICT production typically focus on generating radical innovations. Academic spin-offs and new firm formation are important mechanisms in this respect. Universities, government labs and other research institutions provide scientific research inputs for innovating firms. Regions well endowed with generic factors such as excellent research universities, knowledge mediating institutions, venture capital organizations and highly skilled labour enjoy a better chance to generate successful ICT clusters. Since regions already hosting successful high technology industries typically fall into this category, one can expect a significant path dependence in these processes.

In regions characterized by more traditional regional innovation systems the RIS will have to undergo a far reaching transformation for high technology clusters to emerge. Rebuilding a RIS is a complex and demanding process. The promotion of excellent research and education institutes and of supporting agents like science parks, academic spin-off centres, etc. is just

one aspect of the necessary restructuring. New routines, attitudes and patterns of behaviour must emerge, requiring substantial amounts of learning and unlearning the old routines and attitudes. This is particularly challenging for policy.

As far as the third perspective, RIS and the use of ICT, is concerned, it is argued that the adoption of ICT is strongly related to firms' strategies, to their organizational practices and to their competitive environment. A simple "transplantation" of ICT onto regional economies does not seem a viable strategy. Success depends upon the socio-economic conditions and in addition to hard infrastructure also on the existence of soft infrastructure elements like organizational qualities and innovation attitudes in companies. While the potentials of ICT are in principle available to every region in Europe, evidence shows that the regions differ enormously in the effectiveness with which they exploit these potentials. This "absorption capacity" to exploit ICT effectively is closely related to the availability of the mentioned elements of an entrepreneurial regional innovation system.

It is argued that *ICT also feeds back onto the functioning of the respective RIS*. In recent history it was often the needs of major components of the innovation systems that triggered successful innovations in ICT. We discuss the development of the Internet as a typical example of this relationship and the way in which this important element of ICT can support key tasks in the functioning of a RIS: information; communication and coordination; access to resources; and cooperation and collaboration.

Globalisation and ICT in the Knowledge-based Economy

ICT is a key factor driving economic globalisation and associated economic changes.

Innovation in ICT is a critical component which can reduce both fixed and variable transactions costs of market entry and, more generally, economic performance. However, while the remarkable changes in the capacities of machines to process, communicate and store information have led to enormous efficiency and other gains, the potential downside of such rapid changes has been very rapid rates of physical and human capital depreciation. The size of firms may be a critical determinant of their access to resources, which are essential to investing and risk taking in order to promote ICT adoption.

The ICT is a major driving force for outsourcing and offshoring. The rapid advancement of technology, in particular ICT, and the worldwide deregulation and competition in the

telecommunications industry have led to a substantial decrease in adjustment and communication costs faced by firms. Increased *global connectivity* and the resulting *international cost saving opportunities* for production mean that firms now face decisions to *outsource and offshore* components of their production process.

ICT intensive firms purchase greater amounts of business services on the market. It has been argued that European countries slow take up of ICT is acting as a barrier to fragmentation and specialisation of business processes.

A large proportion of service sector occupations are found to be potentially offshorable. Recent research suggests that the share of exports of business service in GDP, the share of ICT investment in total gross fixed investment, the share of services sector in GDP and human capital are positively associated with an increase in the share of employment in potentially offshorable non-clerical occupations, while the share of imports has an offsetting effect on the share of employment in potentially offshorable non-clerical occupations. Furthermore, an increasing share of employment in potentially offshorable clerical occupations, the exports to GDP ratio, the human capital measure, share of hi-tech output in GDP, and product market regulations are positively related. Imports of business services, declining trade union densities and rising share of services in GDP are negatively related.

From the perspective of *regional economic performance*, globalisation potentially not only offers unique opportunities for regions to reap returns from unique assets on expanded, increasingly worldwide markets, but also permits them escaping the constraints of unfavourable local economic conditions. Reduced international transaction costs, facilitated by the use of ICT can redefine the nature and functioning of markets, their interconnectivity, as well as the relative importance of spatial economic factors. As a result, there are potentially asymmetric changes in certain of the relative costs of agents, which are often interrelated in complex ways with spatially sensitive economic factors and, in turn, agents' competitiveness. For example, while a firm in a remote region, may be able to effectively use e-marketing to sell a unique product, its ability to satisfy customers could be dependent on complementary infrastructure investments which would reduce the transportation costs for the delivery of the goods to potential clients.

By reducing organizational costs of coordination, ICT facilitates the locational dispersion, fragmentation and decentralization. This applies to the exchange of goods and services both within and outside institutions original structures.

Due to the increased pace of technological change, globalisation also entails heightened opportunity costs of poor policy choices and design. The heightened interdependence linked to economic globalisation is redefining the gap between technological frontier countries/regions and economic leaders, on the one hand, and technological followers and poorer countries/regions.

Thus, it appears that proactive country and regional policies favouring innovation and ICT diffusion are increasingly of primordial importance. Effective policies need to support mechanisms that generate a concentration of certain economic activities in order to realize increased scale economies and/or agglomeration effects, which correspond to the internalization of positive location-specific spillover effects (i.e. externalities). In an increasingly knowledge-based economy, worldwide, the latter include R&D spillovers, which can be linked to a variety of factors including the quality of national and regional innovation systems. More specifically, technological competitiveness can be critically impacted by university-industry linkages, the state of human capital development, labour-market synergies linked to complementary training in different companies, etc.

7.2 ICT, Innovation Systems and Regional Development: An Integrated View

In the previous chapters of this paper we have discussed various aspects of the complex relationship between ICT and regional development. We have highlighted the key aspects of ICT as a general purpose technology, discussed the economic impacts of ICT diffusion from a macro as well as from a micro perspective, and discussed the spatial consequences of ICT diffusion. The fairly general concepts of regional innovation systems and of globalization were analysed in order to propose an organizing framework for the mechanisms under discussion.

In this section we attempt to combine these elements into an integrated argument. Given the complexity of the issue that has become apparent in the previous chapters, this integration cannot be made in the form of a consistent and rigorously formulated model, but only in terms of tying together the main arguments.

Although they were made from different conceptual and theoretical perspectives, many of the arguments brought forward in the previous chapters were quite similar and closely related. For example, the role of knowledge and human capital was not only stressed in the discussion of regional innovation systems in Chapter 5, but also in Chapters 3 and 4. Similarly, the interdependence of regional economies and the competition between them, which is a major argument in the context of globalization, is also stressed in chapter 5 where we discussed the interdependence between regional, national, and international innovation systems.

The relationship between ICT and regional development is manifold and complex. In our view the concepts of globalization and (regional) innovation systems are suitable frameworks for the discussion of this relationship. Neither "ICT" nor the "Region" should be viewed as homogeneous entries. Both consist of numerous elements and are in close competition with other regions or other sectors of the economy. This competition ties both regions and ICT sector firms into the processes of globalization: increased capital mobility, rapidly changing environments, increased competition with declining profit margins, etc.

The tendency of ICT to reduce transaction costs and to facilitate communication and control over longer distances is a major factor behind globalization. It opens up new markets and new opportunities for firms, in many cases radically changing quite traditional industries. The respective competitive pressures force companies to reconsider their internal organization as well as their position in the value added chain, which generally becomes more fragmented. The application of ICT allows for new forms of (spatial) specialization leading via outsourcing and offshoring to a new structure of the spatial division of labour. The corresponding processes of capital mobility directly impact regions, their population, economy, and public policy.

All these processes take place in a dynamic rather than in a static economic environment. The above mentioned competitive pressures force companies into a continuous race for the innovative advantage; be it in terms of new products, new production processes, or new forms of organization. As has been discussed above, however, the innovation process is highly complex, influenced by many factors, many of which themselves are influenced by innovation, and highly uncertain and risky in its outcome. The literature on innovation systems argues that the innovation process is of complex dynamics, path dependent and subject to cumulative feedback loops. The availability of specific types of ICT, for example,

is an important prerequisite for innovations in ICT. Such mechanisms can lead to selfsustained growth processes on the one hand, or to lock-in in a technological dead-end on the other.

A key factor in the (regional) innovation systems literature is knowledge. As has been described in chapter 5 above, knowledge can be of very different types. It can be quite general in the form of basic human capital or highly specific and applicable only to a very special task. It can be codified and therefore easily transferable or tacit and thus tied to the individual possessing this knowledge. Different types of knowledge, be it part of a synthetical knowledge base or an analytical knowledge base, are needed for solving different types of problems. Accumulated knowledge from the past may be both, a prerequisite for or a factor hampering the acquisition of new knowledge.

Since regional development in the long run depends upon successful innovations, generation of knowledge and the design of the region's innovation system are key factors for the economic viability of a region. Inventions and new information are necessary, but by no means sufficient for successful innovations. The regional economy and population also need the capacity to absorb the new elements and to utilize them to their full capacity. In this sense, the availability of ICT in a region is no guarantee for its productive use in the regional economy. The region also needs the human capital and the RIS environment in order to absorb the technology and to realize its potential benefits. Many factors like the quality of educational institutions, the availability of venture capital, a positive attitude toward risk taking and innovation, etc. are claimed to be necessary environmental factors.

A major element of the innovation process is the ability and skill to combine information and knowledge from various sources with previous experience. The ability of ICT to reduce transaction costs, ease the exchange of information, and to store and organize codified knowledge makes it an important resource for the generation of new skills and the innovation process in general; in addition to the more traditional factor that the introduction of ICT can improve many products, services, and production processes. In addition to the externalities generated by the spillovers of knowledge from innovative agents to others, the network externalities of the ICT infrastructure contribute to the complexity of the relationship between ICT and regional development.

This complex and highly nonlinear set of relationships between the various aspects of ICT, the many elements of the innovation system and the innovative capacity of a regional economy facing the competitive pressures of globalization, represents a major challenge for policy at the regional, national and European level. In such a complex system there does not exist the one critical policy variable. It typically requires the contributions of most of the factors discusses so far. Because of complementarities lack of one factor may severely hamper or even block the whole system. Because of the inherent path dependence minor differences between regions may set them off at quite different trajectories in terms of ICT adoption, innovative environment, and regional economic prosperity. The interplay of those factors is essential, but hardly observable in aggregate statistics. It needs a case study approach to identify the most critical factors and to observe their complex interrelation.

References

- Abramovsky, L. and Griffith, R., 2005 Outsourcing and offshoring of business services: how important is ICT? IFS Working Papers, W05/22.
- Acs, Z. (Ed.) (2000): Regional Innovation, Knowledge and Global Change. Pinter, London.
- Aghion, P. & P. Howitt (1992), A Model of Growth through Creative Destruction, Econometrica 60, 323-351.
- Ahmed, N., P. Schreyer & A. Wölfl (2004), ICT Investment in OECD Countries and its Economic Impact, in OECD (2004), The Economic Impact of ICT Measurement, Evidence and Implications, OECD, Paris, 61-83.
- Ambos, B., 2005. Foreign direct investment in industrial research and development: A study of German MNCs. Research Policy 34, 4, 395-410.
- Amin, A. and Cohendet, P. (2004): Architectures of knowledge. Oxford University Press, Oxford.
- Amiti, M., Wei, Shang-Jin., (2006) "Service offshoring, productivity and employment: evidence from the US", CEPR Discussion Paper No 5475, www.cepr.org
- Andersson, M. & C. Karlsson (2004), The Role of Accessibility for the Performance of Regional Innovation Systems, in Karlsson, C., P. Flensburg & S.-Å. Hörte (2004) (Eds.), Knowledge Spillovers and Knowledge Management, Edward Elgar, Cheltenham, 283-310.
- Andersson, M. & C. Karlsson (2006), Regional Innovation Systems in Small and Medium-Sized Regions, in Johansson, B., C. Karlsson & R.R. Stough (2006) (Eds.), The Emerging Digital Economy. Entrepreneurship, Clusters and Policy, Springer, Berlin, 55-81.
- Andersson, M., U. Gråsjö & C. Karlsson (2007), Human Capital and Productivity in a Spatial Economic System, Annales d'economie et de statistique (Forthcoming)
- Andersson, Å.E. & D.E. Andersson (2006), The Economics of Experiences, the Arts and Entertainment, Edward Elgar, Cheltenham.
- Anderson, T. (2001): Information technology, strategic decision making approaches and organizational performance in different industrial settings. Journal of Strategic Information Systems, 10, pp. 101-119.
- Audretsch, D.B. & M.P. Feldman (1996), R&D Spillovers and the Geography of Innovation and Production, American Economic Review 86, 630-640.
- Angeriz, A., McCombie, J., Roberts, M. (2006) Some New Estimates of Returns to Scale for EU Regional Manufacturing, 1986-2002, CCEPP, Working Papers, WP03-06.
- Anselin, L., Varga, A. and Acs, Z. (1997): Local Geographic Spillovers between University Research and High Technology Innovations. Journal of Urban Economics, 42 (3), pp. 422-448.
- Antras, Pol (2003), Firms, Contract and Trade Structure, Quarterly Journal of Economics, 118, 1375-1418.
- Antras, Pol and Elhanan Helpman (2004), .Global Sourcing, Journal of Political Economy, Vol. 112, No. 3, June 2004, pp. 552-580.
- Archibugi, D. and Michie, J. (Eds.) (1997): Technology, Globalisation and Economic Performance, Cambridge University Press, Cambridge.
- Archibugi, D., Howells, J. and Michie, J. (Eds.) (1999): Innovation Policy in a Global Economy, Cambridge University Press, Cambridge.
- Asheim, B. and Isaksen, A. (2002): Regional Innovation Systems: The Integration of Local 'Sticky' and Global 'Ubiquitous' Knowledge. Journal of Technology Transfer, 27, pp. 77-86.

- Asheim, B. and Gertler, M. (2005): The Geography of Innovation, in: Fagerberg, J. Mowery, D. and Nelson, R. (Eds.), The Oxford Handbook of Innovation. Oxford University Press, Oxford, pp. 291-317.
- Asheim, B. and Coenen, L. (2006): Contextualising Regional Innovation Systems in a Globalising Learning Economy: On Knowledge Bases and Institutional Frameworks. Journal of Technology Transfer, 31, pp. 163-173.
- Aubert, P., E. Caroli & M. Roger (2006), New Technologies, Organisation and Age: Firm-Level Evidence, The Economic Journal 116, F73-F93.
- Audretsch, D. and Feldman, M. (1996): Innovative Clusters and the Industry Life Cycle. Review of Industrial Organisation, 11, pp. 253-273.
- Autio, E. (1998): Evaluation of RTD in Regional Systems of Innovation. European Planning Studies, 6, pp. 131-140.
- Bailey, M.N. & R. Gordon (1988), The Productivity Slowdown, Measurement Issues and the Explosion of Computer Power, Brookings Papers on Economic Activity, 347-423.
- Baily, M.N. & R. Solow (2001), International Productivity Comparisons Built from the Firm Level, Journal of Economic Perspectives 15, 151-172.
- Baldwin, Robert E, and Cain, Glen G. Trade and U.S. Relative Wages. Mimeo, University of Wisconsin, 1994.
- Baptista, R. & P. Swann (1998), Do Firms in Clusters Innovate More? Research Policy 27, 525-540.
- Bardhan, A. D., and Kroll, C. (2003), 'The new wave of outsourcing', University of California Berkeley, Fisher Centre for Real Estate and Urban Economics, Fisher Centre Research Report No. 1103.
- Barry, F., (2004) Export-platform foreign direct investment: the Irish experience, EIB papers, vol 9 Number 2, European Investment Bank.
- Bartel, Ann, Lach, Saul and Sicherman, Nachum (2005). Outsourcing and Technological Change. NBER Working Paper 11158, February 2005.
- Bathelt, H., Malmberg, A. and Maskell, P. (2004): Clusters and Knowledge: Local Buzz, Global Pipelines and the Process of Knowledge Creation. Progress in Human Geography, 28 (1), pp. 31-56.
- Bellini, E., G.I.P. Ottaviano and D. Pinelli (2003) The ICT Revolution: Opportunities and Risks for the *Mezzogiorno*, FEEM Working Paper No. 86.
- Belorgey, N., R. Lecat & T.-P. Maury (2006), Determinants of Productivity per Employee: An Empirical Estimation Using Panel Data, Economics Letters 91, 153-157.
- Berndt, E.R. & C.J. Morrison (1995), High-Tech Capital Formation and Economic Performance in U.S. Manufacturing Industries: An Exploratory Analysis, Journal of Econometrics 65, 9-53.
- Beyers, W.B. (2000), Cyberspace or Human Space. Wither Cities in the Age of Telecommunications? in Weeler, J.M., Y. Ayoama & B. Warf (2000) (Eds.), Cities in the Telecommunications Age, Routledge, New York, 161-180.
- Bhalla, A., Is more IT offshoring better? An explanatory study of Western companies offshoring to Southeast Asia, Journal of Operational Management 2007-06-25 forthcoming.
- Black, S.E. & L.M. Lynch (2001), How to Compete: The Impact of Workplace Practices and Information Technology on Productivity, Review of Economics and Statistics 83, 434-445.
- Blinder, A. S. (2005), 'Fear of offshoring', Princeton University, CEPS Working Paper No. 119, December 2005.

- Bocquet, R., Brossard, O. and Sabatier, M. (2007): Complementarities in organizational design and the diffusion of information technologies: An empirical analysis. Research Policy, 36, pp. 367-386.
- Bottazzi, L. and Peri, G. (2003): Innovation and spillovers in regions: Evidence from European patent data. European Economic Review, 47, pp. 687-710.
- Boghani, A., E. Kimble & E. Spencer (1991), Can Telecommunications Help Solve America's Transportation Problem? Arthur D. Little, Cambridge, MA.
- Bosworth, B. & J. Triplett (2000), What's New About the New Economy? IT, Economic Growth and Productivity, Brookings Institution, http://www.brook.edu/views/papers/bosworth/20001020.htm
- Bresnaham, T.F., E. Brynjolfson & L.M. Hitt (2002), Information Technology, Workplace Organisation and the Demand for Skilled Labour. Firm-level Evidence, Quarterly Journal of Economics 117, 339-376.
- Bresnahan, T.F. & M. Trajtenberg (1995), General Purpose Technologies: "Engines of Growth", Journal of Econometrics 65, 83-108.
- Bretschek, I. & U. Kaiser (2004), Productivity Effects of Organisational Change: Microeconometric Evidence, Management Science 50, 394-404.
- Bristow, G. (2003), The Implications of the New Economy for Industrial Location, in Jones D.C. (2003) (Ed.), New Economy Handbook, Academic Press, London, 269-287.
- Breschi, S. (2000): The Geography of Innovation: A Cross-Industry Analysis. Regional Studies, 34, pp. 213-229.
- Breschi, S. and Malerba, F. (1997): Sectoral Innovation Systems, Technological Regimes, Schumpeterian Dynamics and Spatial Boundaries, in: Edquist, C. (Eds.), Systems of Innovation. Pinter, London, pp. 130-156.
- Bröcker, J., (1994). Die Lehren der neuen Wachstumstheorie für die Raumentwicklung und die Regionalpolitik. In: U. Blien, H. Herrmann, M. Koller (eds.), Regionalentwicklung und regionale Arbeitsmarktpolitik, Konzepte zur Lösung regionaler Arbeitsmarktprobleme?, Beiträge zur Arbeitsmarkt- und Berufsforschung Nr. 184, Nürnberg: Landesarbeitsamt Nordbayern.
- Brynjolfson, E. & L.M. Hitt (1993), Information Systems Spending Productive? Proceedings of the 14th International Conference on Information Systems.
- Brynjolfson, E. & L.M. Hitt (1996), Paradox Lost? Firm-level Evidence on the Returns to Information Systems Spending, Management Science 42, 541-558.
- Brynjolfson, E. & L.M. Hitt (2000), Beyond Computation: Information Technology, Organisational Transformation and Business Practices, Journal of Economic Perspectives 14, 23-48.
- Brynjolfsson, E. & L.M. Hitt (2003), Computing Productivity: Firm-Level Evidence, Review of Economics and Statistics 85, 793-808.
- Brynjolfson, E., L.M. Hitt & S. Yang (2002), Intangible Assets: Computers and Organisational Capital, Brookings Papers on Economic Activity: Macroeconomics, Vol. 1, 137-199.
- Brynjolfson, E. & B. Kahin (2000), Introduction, in Brynjolfson, E. & B. Kahin (2000) (Eds.), Understanding the Digital Economy. Data, Tools, and Research, The MIT Press, Cambridge, MA, 1-10.
- Buckley, P.J. and Casson, M.C. (1976), The Future of the Multinational Enterprise, Holmes and Meier, London.
- Cairncross, F. (1997), The Death of Distance, Harvard Business School Press, Boston, MA.
- Camagni, R. (1995), The Concept of Innovative Milieu and Its Relevance for Public Policies in European Lagging Regions, Papers in Regional Science 74, 317-340.

- Carlino, G.A. (2001), Knowledge Spillovers: Cities' Role in the New Economy, Federal Reserve Bank of Philadelphia Business Review, Q4, 17-26.
- Carlsson, B. (2003), The New Economy: What is New and what is not? In Christensen, J.F. & P. Maskell (2003) (Eds.), The Industrial Dynamics of the New Digital Economy, Edward Elgar, Cheltenham, 13-32.
- Carlsson, B. (1994): Technological Systems and Economic Performance, in: Dodgson, M., Rothwell, R. (Eds), The Handbook of Industrial Innovation. Edward Elgar, Aldershot, pp. 13-24.
- Castells, M. (1989), The Informational City. Information Technology, Economic Restructuring and the Urban-regional Process, Basil Blackwell, Oxford.
- Castells, M. (1996), The Rise of the Network Society. The Information Age: Economy, Society and Culture. Vol.1, Blackwell, Oxford.
- Clark, C. (1957), Transport: Maker and Breaker of Cities, Town Planning Review 28, 237-250.
- Cohen, D., P. Garibaldi & S. Scarpetta (2004), The ICT Revolution: Productivity Differences and the Digital Divide, Oxford University Press, New York.
- Cohen, G., I. Salomon & P. Nijkamp (2002), Information-communication Technologies (ICT) and Transport: Does Knowledge Underpin Policy? Telecommunications Policy 26, 31-52.
- Cohen-Blankshtain, G. & P. Nijkamp (2004), The Appreciative System of Urban ICT Policies: An Analysis of Perceptions of Urban Policy Makers, Growth and Change 35, 166-197.
- Colecchia, A. & P. Schreyer (2001), The Impact of Information Communications Technologies on Output Growth, STI Working paper 2001/7, OECD, Paris.
- Coleccia, A. & P. Schreyer (2002), The Contribution of Information and Communication Technologies on Economic Growth in Nine OECD Countries, OECD Economic Studies 34.
- COM (2003), European Competitiveness Report 2003, Commission Staff Working Document, European Commission, Brussels.
- COM (2005a), i2010 A European Information Society for Growth and Employment, COM (2005) 229 final, European Commission, Brussels.
- COM (2005b), The Economic Costs of Non-Lisbon. A Survey of the Literature on the Economic Impact of Lisbon-type Reforms, Director-General for Economic and Financial Affairs, No. 16 March, European Commission, Brussels http://europa.eu.int/comm/economy finance/index en.htm
- Cooke, P. (2004): Integrating Global Knowledge Flows for Generative Growth in Scotland: Life Science as a Knowledge Economy Exemplar, in J. Potter (Ed.) Inward Investment, Entrepreneurship and Knowledge Flows in Scotland International Comparisons, OECD, Paris, pp. 73-96.
- Cooke, P., Boekholt, P. and Tödtling, F. (2000): The Governance of Innovation in Europe. Pinter, London.
- Cooke, P. Heidenreich, M. and Braczyk, H.J. Brazcyk (Eds.) (2004): Regional innovation systems (2nd Edition). Routledge, London and New York.
- Cooke, P. (1997), Regions in a Global Market: The Experiences of Wales and Baden-Württemberg, Review of International Political Economy 42, 13-36.
- Cornford, J., Richardson, R., Sokol, M., Marques, P. and Gillespie, A. (2006) Transformation of Regional Societies Through ICT: State(s) of the Art(s), July 2006.[http://europa.eu.int/information_society/eeurope/i2010/docs/i2010_high_level_gr oup/note_on_economic_impact_of_ict.doc].
- Couclelis, H. (1994), Spatial Technologies, Environment and Planning B 21, 142-143.

- Crescenzi, R. (2005), Innovation and Regional Growth in the Enlarged Europe: The Role of Local Innovative Capabilities, Peripherality, and Education, Growth and Change 36, 471-507.
- Crowston, K. & M. Meyers (2004), Information Technology and the Transformation of Industries: Three Research Perspectives, Strategic Information Systems 13, 5-28.
- Daveri, F. (2002), The New Economy in Europe, Oxford Review of Economic Policy 18, 345-362.
- David, P.A. (1991), Computer and Dynamo: The Modern Productivity Paradox in a Distant Mirror, in OECD (1991), Technology and Productivity: The Challenge for Economic Policy, OECD, Paris, 315-348.
- David, P. and Foray, D. (2003): Economic fundamentals of the knowledge society, Policy Futures in Education, 1, pp. 20-49.
- Deardorff, Alan V., (2001), "Fragmentation in Simple Trade Models," North American Journal of Economics and Finance, 12: 2, 121-137.
- Dederick, J., V. Gurbaxani & K.L. Kraemer (2003), Information Technology and Economic Performance: S Critical Review of the Empirical Evidence, ACM Computing Surveys 35, 1-28.
- De la Mothe, J. and Paquet, G. (Eds.) (1998): Local and Regional Systems of Innovation, Kluwer Academic Publishers, Boston.
- Denison, E.F. (1962), The Sources of Economic Growth in the United States and the Alternatives before Us, Committee for Economic Development, Washington, D.C.
- Desrochers, P. (1997), A Geographical Perspective on Austrian Economics, The Quarterly Journal of Austrian Economics 1, 63-83.
- Devaraj, S. & R. Kohli (2000), Information Technology Payoff in the Health-care Industry: A Longitudinal Study, Journal of Management Information Systems 16, 41-67.
- Dixit, A.K., J.E. Stiglitz, (1977). Monopolistic Competition and Optimum Product Diversity, American Economic Review, 67: 297–308.
- Doloreux, D. (2002): What we should know about regional systems of innovation. Technology in Society, 24, pp. 243-263.
- Doloreux, D. and Parto, S. (2005): Regional innovation systems: Current discourse and unresolved issues. Technology in Society, 27 (2), pp. 133-153.
- Dunning, J.H. (1977), Trade, location of economic activity and the multinational enterprise: a search for an eclectic approach. In: B. Ohlin, P. Hesselborn and P. Magnus, Editors, The International Allocation of Economic Activity, Holmes & Meier, New York.
- Easterly, W. & R. Levine (2001), It's Not Factor Accumulation. Stylized Facts and Growth, World Bank Economic Review 15, 177-219.
- Eden, L. and Miller, S.R. (2004), Distance matters: liability of foreignness, institutional distance and ownership, Advances in International Management 16, pp. 187–221.
- Edquist, C. (1997), System of Innovation Approaches Their Emergence and Characteristics, in Edquist, C. (1997) (Ed.), Systems of Innovation: Technologies, Institutions and Organisations, Pinter Publishers/Cassell Academic, London, 343-381.
- Edquist, C. (2001): The Systems of Innovation Approach and Innovation Policy: An account of the state of the art, Lead paper presented at the DRUID Conference, June 12-15, 2001, Aalborg.
- Edquist, C. (2005): Systems of Innovation Perspectives and Challenges, in: Fagerberg, J., Mowery, D., Nelson, R. (Eds.), The Oxford Handbook of Innovation. Oxford University Press, Oxford, pp. 181-208.
- Edquist, C. and Johnson, B. (1997): Institutions and Organizations in Systems of Innovation, in: Edquist, C. (Ed.), Systems of Innovation. Pinter, London, pp. 41-63.

- Ekholm, K., Hakkala, K., (2006) The effect of offshoring on labour demand: evidence from Sweden, CEPR Discussion Paper No. 5648, www.cepr.org
- Ellram, L., Tate, W., Billington, C., Offshore outsourcing of professional services, A transaction cost economics perspective, Journal of Operational management 2007-06-25, forthcoming.
- Fagerberg, J., et al., (2000) (Eds.), The Economic Challenge for Europe, Edward Elgar, Cheltenham.
- Feenstra, R.C., and Hanson, G. H., (1996) Globalisation, Outsourcing, and Wage Inequality, American Economic Review, LXXXVI (1996), 240-245.
- Feenstra, Robert C. & Hanson, Gordon H., (1997). Foreign direct investment and relative wages: Evidence from Mexico's maquiladoras, Journal of International Economics, Elsevier, vol. 42(3-4), pages 371-393, May.
- Feldman, M.P. & D. Audretsch (1999), Innovation in Cities: Science-Based Diversity, Specialisation and Localised Competition, European Economic Review 43,409-429.
- Fifarek, B., Veloso, F., Davidson, C., Offshoring technology innovation: A case study of rare Earth technology, Journal of Operational Management 2007-06-25.
- Flamm, K. (1999), Digital Convergence? The Set-top Box and the Network Computer, in Eisenach, J.A. & T.M. Lenard (1999) (Eds.), Competition, Innovation and the Microsoft Monopoly: Antitrust in the Digital Market Place, Kluwer Academic Publishers, Boston, 255-290.
- Fornahl, D. and Brenner, T. (Eds.) (2003), Cooperation, Networks and Institutions in Regional Innovation Systems, Edward Elgar, Cheltenham.
- Franke, R.H. (1987), Technological Revolution and Productivity Decline: Computer Introduction in the Financial Industry, Technological Forecasting and Technological Change 31, 143-154.
- Freeman, C. (1987), Technology Policy and Economic Performance: Lessons from Japan, Pinter, London.
- Freeman, C. & C. Perez (1988), Structural Crises of Adjustment: Business Cycles and Investment Behaviour, in Dosi, G. et al. (1988) (Eds.), Technical Change and Economic Theory, Pinter, London.
- Friedman, T.L. (2005), The World is Flat. A Brief History of the Twenty-First Century, Farrar, Strauss and Giroux, New York.
- Fujita, M., P. Krugman & A. Venables (1999), The Spatial Economy. Cities, Regions and International Trade, The MIT Press, Cambridge, MA.
- Garrison, W. & E. Deakin (1988), Travel, Work, and Telecommunications: A Long View of the Electronic Revolution and Its Potential Impact, Transportation Research 22A, 2390-2246
- Gaspar, J. & E.L. Glaeser (1998), Information Technology and the Future of Cities, Journal of Urban Economics 43, 136-156.
- Gersbach, H., Schmutzler, A., 2006. Foreign direct investment and R&D offshoring. Working Paper, University of Zurich.
- Gertler, M. (2003): Tacit knowledge and the economic geography of context or The undefinable tacitness of being (there). Journal of Economic Geography, 3, pp. 75-99.
- Gertler, M. and Levitte, Y. (2005): Local Nodes in Global Networks: The Geography of Knowledge Flows in Biotechnology Innovation. Industry and Innovation, 12(4), pp. 487-507.
- Gertler, M. and Wolfe, D. (2005): Spaces of Knowledge Flows: Clusters in a Global Context. Paper presented at the DRUID Tenth Anniversary Summer Conference on "Dynamics of Industry and Innovation: Organizations, Networks and Systems", Copenhagen, June 27-29, 2005.

- Giannopoulus, G. & A. Gillespie (1993) (Eds.), Transport and Communications Innovation in Europe, Belhaven, London.
- Glaeser, E.L., et al. (1992), Growth in Cities, Journal of Political Economy 100, 1126-1152
- Goddard, J. (1991), New Technology and the Geography of the UK Information Economy, in Brotchie, J. et al. (1991) (Eds.), Cities of the 21st Century. New Technologies and Spatial Systems, Halsted Press, New York, 191-213.
- Gordon, R.J. (2000), Does the 'New Economy' Measure Up to the Great Inventions of the Past? Journal of Economic Perspectives 14, 49-74.
- Gordon, R.J. (2004), Why Was Europe Left at the Station when America's Productivity Locomotive Departed? NBER Working Paper Series 10661.
- Gorg, H.; Hanley, A.(2004)., Does Outsourcing Increase Profitability? Economic and Social Review, Vol.35 (3), pp. 367-387.
- Graham, S. (2002), Bridging Urban Digital Divides? Polarisation and Information and Communication Technologies (ICT), Urban Studies 39, 33-56.
- Graham, S. & S. Marvin (1996), Telecommunications and the City. Electronic Spaces, Urban Places, Routledge, New York.
- Graham, S. & S. Marvin (2000), Urban Planning and the Technological Future of Cities, in Weeler, J.M., Y. Ayoama & B. Warf (2000) (Eds.), Cities in the Telecommunications Age, Routledge, New York, 71-96.
- Grant, A.E. & L. Berqiust (2000), Telecommunications Infrastructure and the City: Adapting to the Convergence of Technology and Policy, in Weeler, J.M., Y. Ayoama & B. Warf (2000) (Eds.), Cities in the Telecommunications Age, Routledge, New York, 97-111.
- Groenewegen, J. and van der Steen, M. (2006): The Evolution of National Innovation Systems. Journal of Economic Issues, XL (2), pp. 277-285.
- Grossman, G. and Rossi-Hansberg, E., (2006), Trading Tasks: A Simple Theory of Offshoring, NBER Working Paper No. W12799 http://www.nber.org/papers/w12721
- Grossman, G.M. & E. Helpman, (1991), Innovation and Growth in the Global Economy, The MIT Press, Cambridge, MA.
- Grossman, Gene and Elhanan Helpman (2002), Integration versus outsourcing in industry equilibrium, Quarterly Journal of Economics 117: 85-120, 2002.
- Grossman, Gene and Elhanan Helpman (2005), Outsourcing in a global economy, Review of Economic Studies 72: 135-159.
- Grossman, Sanford J. and Oliver D. Hart (1986), "The Costs and Benefits of Ownership: A Theory of Vertical and Lateral Integration," Journal of Political Economy, 94:4, pp. 691-719.
- Grunwald, J. & K. Flamm (1985), The Global Factory: Foreign Assembly in International Trade, Brookings Institution Press, Washington, D.C.
- Gust, C. & J. Marquez (2004), International Comparisons of Productivity Growth: The Role of Information Technology and Regulatory Practices, Labour Economics 11, 33-58.
- Hall, P. (1998), Cities in Civilization. Culture, Innovation and Urban Order, Weinfield and Nicholson, London.
- Hart, Oliver and John Moore (1990), Property Rights and the Nature of the Firm, Journal of Political Economy, 98:6, pp. 1119-1158.
- Hekman, J.S. (1982), What are businesses looking for? Federal Reserve Bank of Atlanta Economic Review 67, pp. 6–19.
- Helpman, Elhanan and Paul R. Krugman (1985), Market Structure and Foreign Trade, Cambridge: MIT Press.
- Helpman, E. & M. Trajtenberg (1998), A Tie to Sow and a Time to Reap: Growth Based on General Purpose Technologies, in Helpman, E. (1998) (Ed.), *General Purpose Technologies and Economic Growth*, The MIT Press, Cambridge MA, 55-83.

- Hepworth, M., & K. Ducatel (1992), Transport in the Information Age: Wheels and Wires, Belhaven, London.
- Hirschman, A.O., (1958). The Strategy of Economic Development, New Haven, Conn.: Yale University Press.
- Hollanders, H. (2007): 2006 European Regional Innovation Scoreboard (2006 RIS), MERIT, Maastricht.
- Howells, J. (1999), Regional Systems of Innovation?, in Archibugi, D., J. Howells & J. Michie (1999), Innovation Policy in a Global Economy, Cambridge University Press, Cambridge, 67-93.
- Howells, J. (2002): Tacit knowledge, innovation and economic geography, Urban Studies, 39, pp. 871-884.
- Howitt, P. (1998), Measurement, Obsolescence, and General Purpose Technologies, in Helpman (1998) (Ed.), General Purpose Technologies and Economic Growth, The MIT Press, Cambridge, MA, 219-251.
- Hymer, S.H. (1976), The International Operations of National Firms: A Study of Direct Investment, MIT Press, Cambridge.
- I2010 High Level Group (2006): The economic impact of ICT: evidence and questions, April 20.
- Jacobs, J. (1969), The Economy of Cities, Random House, New York.
- Jaffe, A., M. Trajtenberg & R. Henderson (1993), Geographical Localisation of Knowledge Spillovers as Evidenced by Patent Citations, Quarterly Journal of Economics 108, 577-598.
- Jaffe, A. (1989): The real effects of academic research. American Economic Review, 79 (5), pp. 957-970.
- Jalava, J. & M. Pohjola (2002), Economic Growth in the New Economy: Evidence from Advanced Countries, Information Economics and Policy 14, 189-210.
- Jensen, J.B., and Kletzer, L. (2005), 'Tradable services: Understanding the scope and impact of services offshoring', Institute for International Economics Working Paper No WP05-9, September 2005.
- Johnson, B. (1992): Institutional Learning, in: Lundvall, B.-A. (Ed.), National Systems of Innovation. Towards a Theory of Innovation and Interactive Learning. Pinter Publishers, London, pp. 23-44.
- Johansson, B. (2006), Spatial Clusters of ICT Industries, in Johansson, B., C. Karlsson & R.R. Stough (2006) (Eds.), The Emerging Digital Economy. Entrepreneurship, Clusters and Policy, Springer, Berlin, 137-167.
- Johansson, B. & C. Karlsson (2001), Geographic Transaction Costs and Specialisation Opportunities of Small and Medium-Sized Regions: Scale Economies and Market Extension, in Johansson, B., C. Karlsson & R.R. Stough (2001) (Eds.), Theories of Endogenous Regional Growth. Lessons for Regional Policies, Springer, Berlin, 150-180.
- Johansson, B., C. Karlsson & R.R. Stough (2002) (Eds.), Regional Policies and Comparative Advantage, Edward Elgar, Cheltenham.
- Johansson, B., C. Karlsson & R.R. Stough (2006), Entrepreneurship, Clusters and Policy in the Emerging Digital Economy, in Johansson, B., C. Karlsson & R.R. Stough (2006) (Eds.), The Emerging Digital Economy. Entrepreneurship, Clusters and Policy, Springer, Berlin, 1-19.
- Jorgensen, D.W. (1963), Capital Theory and Investment Behaviour, American Economic Review 53, 247-259.
- Jorgensen, D.W. (2001), Information Technology and the U.S. Economy, American Economic Review 91, 1-32.

- Jorgensen, D.W. & Z. Griliches (1967), The Explanation of Productivity Change, Review of Economic Studies 34.
- Jorgensen, D.W., Ho, M.S. & K.J. Stiroh (2002), Projecting Productivity Growth: Lesson from the US Growth Resurgence, Federal Reserve Bank of Atlanta Economic Review, Third Quarter, 1-13.
- Jorgensen, D.W. & K. Stiroh (1999), Information Technology and Growth, American Economic Review 89, 109-116.
- Kaufmann, A., Lehner, P. and Tödtling, F. (2003) Effects of the Internet on the spatial structure of innovation networks. Information Economics and Policy, 15, pp. 402-424.
- Karlsson, C. & M. Andersson (2007), Knowledge in Regional Economic Growth The Role of Knowledge Accessibility, Industry and Innovation (Forthcoming).
- Karlsson, C. & B. Johansson (2006), Dynamics and Entrepreneurship in a Knowledge-based Economy, in Karlsson, C., B. Johansson & R.R. Stough (2006) (Eds.), Entrepreneurship and Dynamics in the Knowledge Economy, Routledge, New York, 12-46.
- Kaufmann, A., Lehner, P. and Tödtling, F. (2003) Effects of the Internet on the spatial structure of innovation networks. Information Economics and Policy, 15, pp. 402-424.
- Keeble, D. and Wilkinson, F. (Eds.) (2000): High-Technology Clusters, Networking and Collective Learning in Europe. Ashgate, Aldershot.
- Kirkegaard, J. F. (2004a), Outsourcing stains on the white collar? Institute for International Economics, Washington, D.C.
- Kirkegaard, J. F. (2004b), Offshore Outsourcing Much Ado about What? CESinfo Forum 5, 22-29.
- Knoke, K. (1996), Bold New World: The Essential Road Map to the Twenty-first Century, Kodansha, New York.
- Kohler, Wilhelm, (2004), "Aspects of International Fragmentation," Review of International Economics, 12:5, 793-816.
- Kohli, R. & S. Devaraj (2003), Measuring Information Technology payoff: A Meta Analysis of Structural Variables in Firm-level Empirical Research, Information Systems Research 14, 127-145.
- Kolko, J. (1999), The Death of Cities? The Death of Distance? Evidence form the Geography of Commercial Internet Usage, paper presented at the workshop "Cities in the Global Information Society: An International Perspective", 22-24 November, Newcastle-upon-Tyne.
- Kolko, J. (2002), Silicon mountains, Silicon Molehills: Geographic Concentration and Convergence of Internet Industries in the US, Information Economics and Policy 14, 211-232.
- Koski, H., Rouvinen, P., and Ylä-Anttila, P. (2002): ICT clusters in Europe: The Great central Banana and the Small Nordic Potato, Information Economics and Policy, 14, pp. 145-165.
- Kuemmerle, W., 1999. The drivers of foreign direct investment into research and development: an empirical investigation. Journal of International Business Studies 30, 1-24
- Krugman, P. (1991), Geography and Trade, The MIT Press, Cambridge, MA.
- Krugman, P.R., (1991b) Increasing Returns and Economic Geography, Journal of Political Economy, 99: 483-499.
- Kuznets, S. (1959), Six Lectures on Economic Growth, Free Press, New York.
- Le Bas, C., Sierra, C., 2002. Location versus home country advantages' in R&D activities: some further results on multinationals' locational strategies. Research Policy 31, 589-609.

- Leamer, E. & M. Storper (2001), The Economic Geography of the Internet, Journal of International Business Studies 32, 641-665.
- Lee, H.G. & T. Clark (1997), Market Process Reengineering through Electronic Market Systems: Opportunities and Challenges, Journal of Management Information Systems 13, 13-30.
- Leiner, B.M., Cerf, V.G., Clark, D.D., Kahn, R.E., Kleinrock, L., Lynch, D.C., Postel, J., Roberts, L.G., Wolff, S. (2003): A brief history of the Internet, http://www.isoc.org/internet/history/brief.shtml
- Lera-López, F. & M. Billón-Currás (2005), Shortfalls and Inequalities in the Development of E-Economy in the E.U.-15, Communications & Strategies 60, 181-200.
- Li, Y., Liu, Y., Li, M., and Wu, H., Transformational offshore outsourcing: empirical evidence from alliances in China, Journal of Operational Management 2007-06-25 forthcoming.
- Lipsey, R.G. C. Becar & K. Carlaw (1998), What Requires Explanation? in Helpman, E. (1998) (Ed.), General Purpose Technologies and Economic Growth, The MIT Press, Cambridge, MA, 15-54.
- Luo, Y., Shenkar, O. and Nyaw, M.-K. (2002), Mitigating liabilities of foreignness: defensive versus offensive approaches, Journal of International Management 8 (3), pp. 283–300.
- Lucas, R.E. (1988), On the Mechanics of Economic Development, Journal of Monetary Economics 22, 3-42.
- Lundvall, B.-Å. (1992), National Systems of Innovation: Towards a Theory of Innovation and Interactive Learning, Pinter, London.
- Lundvall, B.-A. (2007): National Innovation Systems Analytical Concept and Development Tool. Industry and Innovation, 14, pp. 95-119.
- Lösch, A. (1954), The Economics of Location, Yale University Press, New Haven.
- Maier, G., Trippl, M. (2007): Location/Allocation of Regional Growth, in: Capello, R., Nijkamp, P. (eds.), Advances in Regional Economics, Celtenham: Edward Elgar.
- Mairesse, J., G. Cette & Y. Kocoglu (2000), Les technologies de l'information et de la communication en France: diffusione t contribution á la crossance, Économie et Statistique 9/10, 117-146.
- Malecki, E.J. (1991), Technology and Economic Development, Longman, New York.
- Malecki, E.J. & S.P. Gorman (2001), Maybe the Death of Distance, but not the End of Geography, in Leinbach, T.R. & S.D. Brunn (2001) (Eds.), Worlds of E-commerce: Economic, Geographical and Social Dimensions, Wiley, Chichester, 87-105.
- Malerba, F. (2002): Sectoral systems of innovation and production. Research Policy, 31, pp. 247-264.
- Malmberg, A. and Maskell, P. (2002): The elusive concept of localization economies: towards a knowledge-based theory of spatial clustering. Environment and Planning A, 34 (3), pp. 429-449.
- Malmberg, A. and Maskell, P. (2006): Localized Learning Revisited. Growth & Change, 37 (1), pp. 1-18.
- Malone, T., J. Yates & R. Benjamin (1987), Electronic Markets and Hierarchies, Communication of the ACM 30, 484-497.
- Mankiw, N. G., and Swagel P. (2005), 'The economics and politics of offshore outsourcing', American Enterprise Institute for Public Policy Research Working Paper No.122, December 2005.
- Marin, D. and Verdier. T., (2005), Corporate Hierarchies and International Trade: Theory and Evidence. PSE, mimeo.
- Marshall, A. (1920), Principles of Economcis, 8th ed., Macmillan, London.

- Maskell,P., Bathelt,H. and Malmberg,A. (2004): Temporary clusters and knowledge creation: the effects of international trade fairs, conventions and other professional gatherings. SPACES Spatial Aspects Concerning Economic Structures, 4, Marburg: Faculty of Geography, Philipps-University of Marburg.
- Maurseth, P.B. & B. Verspagen (1999), Europe: One or Several Systems of Innovation? An Analysis Based on Patent Citations, in Fagerberg, J., P. Guerrieri & B. Verspagen (1999) (Eds.), The Economic Challenge for Europe, Edward Elgar, Cheltenham, 18-43.
- Meijers, H. (2004), Productivity in Europe and the U.S.: ICT and the Role of Network Effects, Communications & Strategies, 1-29.
- Metters, R., The history of offshoring knowledge services, Journal of Operational management 2007-06-25, forthcoming.
- Metters, R., A typology of offshoring and outsourcing electronically transmitted services, Journal of Operational management 2007-06-25 forthcoming.
- Mokhtarian, P. (1991), Telecommuting and Travel: State of the Practice, State of the Art, Transportation 18, 319-342.
- Mokhtarian, P. & R. Meenaksisundaran (1999), Beyond Tele-substitution: Disaggregated Longitudinal Structural Equation Modelling of Communication Impacts, Transportation Research Part C 7, 33-52.
- Mokhtarian, P. & I. Salomon (2001), How Derived Is Demand for Travel? Some Conceptual and Measurement Considerations, Transportation Research A: Policy and Practice 35, 695-719.
- Morgan, K. (2004): The exaggerated death of geography: learning, proximity and territorial innovation systems. Journal of Economic Geography, 4, pp. 3-21.
- Morrison, C.J. & E.R. Berndt (1991), Assessing the Productivity Impact of Information Technology Equipment in the U.S. Manufacturing Industries, NBER Working Paper #3582.
- Moss, M. (1991), The Information City in the Global Economy, in Brotchie, J., et al. (1991) (Eds.), Cities of the 21st Century. New Technologies and Spatial Systems, Halsted, New York, 181-190.
- Mowery, D.C. & T. Simcoe (2002), Is the Internet a US Invention? An Economic and Technological History of Computer Networking, Research Policy 31, 1369-1387.
- Mudambi, R. (1995), The MNE investment location decision: some empirical evidence, Managerial and Decision Economics 16, pp. 249–257.
- Myrdal, G., (1957). Economic Theory and Underdeveloped Regions, London: Duckworth.
- Naisbitt, R. (1995), The Global Paradox, Avon Books, New York.
- Naghavi, A. and G. Ottaviano (2006) "Offshoring and Product Innovation", CEPR Discussion Paper No. 6008, www.cepr.org
- Negroponte, N. (1995), Being Digital, Vintage Books, New York.
- Nelson, R. (Ed.) (1993): National Innovation Systems A Comparative Analysis. Oxford University Press, Oxford.
- Nijkamp, P., G. Pepping & D. Banister (1995), Telematics and Transport Behaviour, Springer, Berlin.
- Nijkamp, P., G. Linders & H.L.F. de Groot (2002), Distance Matters, in Poot, J. (2002), The New Information Society and the Periphery, Edward Elgar, Cheltenham.
- Nilles, J.M. (1988), Traffic Reduction by Telecommuting: A Status Review and Selected Bibliography, Transportation Review 22A, 301-317.
- OECD (1999): Managing National Innovation Systems. OECD, Paris.
- OECD (2002), Measuring the Information Economy. The ICT Sector, in OECD Proceedings Science and Information Technology, 35-69.

- OECD (2003), ICT and Economic Growth: Evidence from OECD Countries, Industries and Firms, OECD, Paris.
- OECD (2004), The Economic Impact of ICT. Measurement, Evidence and Implications, OECD, Paris.
- Ogawa, H. (2000), Spatial Impact of Information Technology Development, The Annals of Regional Science 34, 537-551.
- Oliner, S. & D. Sichel (1994), Computers and Output Growth Revisited: How Big Is the Puzzle?, Brookings Papers on Economic Activity, 273-317.
- O'Mahony, M., & B. van Ark (2003) (Eds.), EU Productivity and Competitiveness: An Industry Perspective. Can Europe Resume the Catching Up Process? Office for Official Publications of the European Communities, Luxembourg.
- Ottaviano, G.I.P, Puga, D. (1997) Agglomeration in the global economy: A survey of the 'new economic geography', Centre for Economic Performance, Discussion Paper No. 356.
- Oulton, N. & S. Srinivasan (2005), Productivity Growth and the Role of ICT in the United Kingdom: An Industry View, 1970-2000, CEP Discussion Paper, No 681, Centre for Economic Performance, London School of Economics, London.
- Owen, Robert F., (2002) "Sunk Costs, Market Access, Economic Integration, and Welfare", (co-authored with David Ulph), *Review of International Economics*, 10(3), pp. 539-555.
- Owen, Robert F., (2007) "Irreversibility, Systemic Endogenous Sunk Costs, News and Evolutionary Economic Methodology", unpublished manuscript, University of Nantes, France, presented at the Western Economic Association Meetings, Seattle.
- Paci, R. and Usai, S. (2000): Technological Enclaves and Industrial Districts: An Analysis of the Regional Distribution of Innovative Activity in Europe. Regional Studies, 34, pp. 97-114.
- Papaconstantinou, G., N. Sakurai & A. Wyckoff (1996), Embodied Technology Diffusion: An Empirical Analysis of 10 OECD Countries, *STI Working Paper 1996/1*, OECD, Paris.
- Polanyi, M. (1966): The Tacit Dimension, Routledge, London.
- Porter, M. (1998): On Competition. Harvard Business School Press, Boston.
- Roach, S.S. (1991), Services under Siege: The Restructuring Imperative, Harvard Business Review 65, 82-91.
- Romer, P.M. (1986), Increasing Returns and Long-run Economic Growth, Journal of Political Economy 94, 1002-1037.
- Romer, P.M. (1990), Endogenous Technological Change, Journal of Political Economy 98, 5-21.
- Rossi-Hansberg, E., Antras, Pol., Garicano, L., Offshoring in a Knowledge Economy, The Quarterly Journal of Economics Feb 2006.
- Sakai, Kentaro (2002), Global Industrial Restructuring: Implications for Small Firms, STI Working Papers, 2002/4, OECD, Paris: OECD.
- Salomon, I. (1986), Telecommunications and Travel Relationships: A Review, Transportation Research A 20A, 223-238.
- Salomon, I. (1998), Technological Change and Social Forecasting: The Case of Telecommuting as a Travel Substitute, Transport Research Part C 6, 17-45.
- Saxenian, A. (1994) Regional Advantage: Culture and Competition in Silicon Valley and Route 128. Harvard University Press, Cambridge, Mass.
- Schemenner, R.W. (1979), Look beyond the obvious in plant location, Harvard Business Review 57, pp. 126–132.

- Schreyer, P., P.-E. Bignon & J. Dupont (2003), OECD Capital Services Estimates: Methodology and a First Set of Results, OECD Statistics Working Paper 2003/6, OECD, Paris.
- Schumpeter, J. (1939), Business Cycles. A theoretical, historical and statistical analysis of the capitalist process, McGraw-Hill, New York.
- Schwartz, G.G. (1990), Telecommunications and Economic Development Policy, Economic Development Quarterly 4, 83-91.
- Sichel, D. (1997), The Computer Revolution. An Economic Perspective, Brookings Institution, Washington, D.C.
- Siegel, D. & Z. Griliches (1991), Purchased Services, Outsourcing, Computers and Productivity in Manufacturing, NBER Working Paper #3678.
- Smith, K. (2002), Assessing the Economic Impacts of ICT, STEP Report R-01, STEP Group, Oslo.
- Soete, L. (2006), Information and Communication Technologies and the New Regional Economy, in Johansson, B., C. Karlsson & R.R. Stough (2006) (Eds.), The Emerging Digital Economy. Entrepreneurship, Clusters and Policy, Springer, Berlin, 21-32.
- Solow, R. (1957), Technical Change and the Aggregate Production Function, Review of Economics and Statistics 39, 312-320.
- Soskice, D. (1999): Divergent Production Regimes: Coordinated and Uncoordinated Market Economies in the 1980s and 1990s, in H. Kitschelt, P. Lange, Marks, G. and Stephens, J. (Eds.), Continuity and Change in Contemporary Capitalism, Cambridge University Press, Cambridge, pp. 101-134.
- Stigler, G. (1951), The Division of Labour is Limited by the Extent of the Market, Journal of Political Economy 59, 185-193.
- Stiroh, K.J. (2002a), Are ICT Spillovers Driving the New Economy? Review of Income and Wealth 48, 33-58.
- Stiroh, K.J. (2002b), Information Technology and the U.S. Productivity Revival: What Do the Industry Data Say? American Economic Review 92, 1559-1576.
- Stiroh, K.J. (2003), Reassessing the Role of IT in the Production Function: A Meta-Analysis http://www.neyyorkfed.org/research/economists/stiroh/ks meta.pdf
- Storper, M. (1997): The Regional World. The Guilford Press, New York.
- Stratman, J., Facilitating offshoring with the enterprise technologies: Reducing operational friction in the governance and production of services, Journal of Operational Management 2007-06-25, forthcoming.
- Swann, G., Prevezer, M. and Stout, D. (1998) (Eds.): The Dynamics of Industrial Clustering: International Comparisons in Computing and Biotechnology, Oxford University Press, Oxford
- Thursby, J., Thursby, M., 2006. Here or there? A survey of factors in multinational R&D location. Report to the government-university-industry research roundtable.
- Tödtling, F. and Trippl, M. (2005) One size fits all? Towards a differentiated regional innovation policy approach. Research Policy, 34, pp. 1203-1219.
- Tödtling, F. and Trippl, M. (2007a): Developing biotechnology clusters in non-high technology regions The case of Austria. Industry & Innovation, 14 (1).
- Tödtling, F. and Trippl, M. (2007b): Knowledge Links in High-Technology Industries: Markets, Networks, or Milieu? The Case of the Vienna Biotechnology Cluster. International Journal of Entrepreneurship and Innovation Management (forthcoming).
- Tödtling, F., Lehner, P. and Trippl, M. (2006) Innovation in knowledge intensive industries: The nature and geography of knowledge links, European Planning Studies, 14 (8), pp. 1043-1066. Toffler, A. (1980), The Third Wave, Bantam Books, New York.

- Traxler, J. & M.I. Luger (2000), Business and Internet: Implications for Firm Location and Clustering, Journal of Comparative Policy Analysis and Practice 2, 279-300.
- Triplett, J.E. & B.B. Bosworth (2002), Baumol's Disease Has Been Cured: IT and Multifactor Productivity in US Service Industries, The Brookings Workshop on Services Industry Productivity, Washington, D.C.
- van Ark, B. (2002), Measuring the New Economy: An International Comparative Perspective, Review of Income and Wealth 48, 1-14.
- van Ark, B., et al. (2003), ICT Investments and Growth Accounts for the European Union, 1980-2000, Research Memorandum GD-56, Groningen Growth and Development Centre, Groningen, http://www.eco.rug.nl/ggdc/homeggdc.html.
- van Ark, B. & M. Piatkowski (2004), Productivity, Innovation and ICT in Old and New Europe, International Economics and Economic Policy 1, 215-246.
- van Welsum, Desiree and Reif, Xavier, (December 2006) "We Can Work it Out the Globalisation of ICT-Enabled Services". NBER Working Paper No. W12799.
- van Welsum, D., and Vickery, G. (2005), 'Potential offshoring of ICT-intensive using occupations', DSTI Information Economy Working Paper, DSTI/ICCP/IE(2004)19/FINAL, OECD, Paris; available at: http://www.oecd.org/sti/offshoring
- Venables, A.J. (2006). Economic Geography, in B.R. Weingast, D. Wittman, The Oxford Handbook of Political Economy, Oxford University Press, 739-754.
- Verspagen, B. (2004), Structural Change and Technology. A Long View, Revue économique 55, 1099-1126.
- Westlund, H. (2006), Social Capital in the Knowledge Economy. Theory and Empirics, Springer, Berlin.
- Wheeler, J.O., Y. Aoyama & B. Warf (2000) (Eds.), Cities in the Telecommunications Age. The Fracturing of Geography, Routledge, New York.
- Whitley, R. (1999): Divergent Capitalism The Social Structuring and Change of Business Systems, Oxford University Press, Oxford.
- Wilson, D. (1995), IT Investment and its Productivity Effects. An Organisational Sociologist's Perspective on Directions for Future Research, Economics of Innovation and New Technology 3, 235-251.
- World Trade Report (2005), http://www.wto.org/english/res_e/booksp_e/anrep_e/world_trade_report05_e.pdf
- W3C, World Wide Web Consortium (2000): A Little History of the World Wide Web, http://www.w3.org/History.html
- Yang, S. & E. Brynjolffson (2001), Intangible Assets and Growth Accounting: Evidence from Computer Investments, www.ebusiness.mit.edu/erik
- Zaheer, S., (1995) Overcoming the liability of foreignness, Academy of Management Journal 38 (2), pp. 341–363.
- Zook, M. (2000), The Web of Production: The Economic Geography of Commercial Internet Content Production in the United States, Environment and Planning A 32, 411-426.
- Zook, M. (2002), Ground Capital: Venture Financing and the Geography of the Internet Industry, 1994-2000, Journal of Economic Geography 2, 151-177.

-		
Year	Number	Title/Author(s) ESRI Authors/Co-authors Italicised
2008	232	Welfare and Competition Effects of Electricity Interconnection between Great Britain and Ireland Laura Malaguzzi Valeri
	231	Is FDI into China Crowding Out the FDI into the European Union? Laura Resmini and <i>Iulia Siedschlag</i>
	230	Estimating the Economic Cost of Disability in Ireland John Cullinan, Brenda Gannon and Seán Lyons
	229	Controlling the Cost of Controlling the Climate: The Irish Government's Climate Change Strategy Colm McCarthy, <i>Sue Scott</i>
	228	The Impact of Climate Change on the Balanced-Growth-Equivalent: An Application of <i>FUND</i> David Anthoff, <i>Richard S.J. Tol</i>
	227	Changing Returns to Education During a Boom? The Case of Ireland Seamus McGuinness, Frances McGinnity, Philip O'Connell
	226	'New' and 'Old' Social Risks: Life Cycle and Social Class Perspectives on Social Exclusion in Ireland Christopher T. Whelan and Bertrand Maître
	225	The Climate Preferences of Irish Tourists by Purpose of Travel Seán Lyons, Karen Mayor and Richard S.J. Tol
	224	A Hirsch Measure for the Quality of Research Supervision, and an Illustration with Trade Economists Frances P. Ruane and Richard S.J. Tol
	223	Environmental Accounts for the Republic of Ireland: 1990-2005 Seán Lyons, Karen Mayor and Richard S.J. Tol
2007	222	Assessing Vulnerability of Selected Sectors under Environmental Tax Reform: The issue of pricing power J. Fitz Gerald, M. Keeney and S. Scott
	221	Climate Policy Versus Development Aid Richard S.J. Tol

220	Exports and Productivity – Comparable Evidence for 14 Countries The International Study Group on Exports and Productivity
219	Energy-Using Appliances and Energy-Saving Features: Determinants of Ownership in Ireland Joe O'Doherty, Seán Lyons and Richard S.J. Tol
218	The Public/Private Mix in Irish Acute Public Hospitals: Trends and Implications Jacqueline O'Reilly and Miriam M. Wiley
217	Regret About the Timing of First Sexual Intercourse: The Role of Age and Context <i>Richard Layte</i> , Hannah McGee
216	Determinants of Water Connection Type and Ownership of Water-Using Appliances in Ireland Joe O'Doherty, Seán Lyons and Richard S.J. Tol
215	Unemployment – Stage or Stigma? Being Unemployed During an Economic Boom Emer Smyth
214	The Value of Lost Load Richard S.J. Tol
213	Adolescents' Educational Attainment and School Experiences in Contemporary Ireland <i>Merike Darmody, Selina McCoy, Emer Smyth</i>
212	Acting Up or Opting Out? Truancy in Irish Secondary Schools Merike Darmody, Emer Smyth and Selina McCoy
211	Where do MNEs Expand Production: Location Choices of the Pharmaceutical Industry in Europe after 1992 Frances P. Ruane, Xiaoheng Zhang
210	Holiday Destinations: Understanding the Travel Choices of Irish Tourists Seán Lyons, Karen Mayor and Richard S.J. Tol
209	The Effectiveness of Competition Policy and the Price-Cost Margin: Evidence from Panel Data Patrick McCloughan, <i>Seán Lyons</i> and William Batt

208	Tax Structure and Female Labour Market Participation: Evidence from Ireland <i>Tim Callan,</i> A. Van Soest, <i>J.R. Walsh</i>
207	Distributional Effects of Public Education Transfers in Seven European Countries <i>Tim Callan,</i> Tim Smeeding and Panos Tsakloglou

European Commission

EUR 24510 EN - Joint Research Centre - Institute for Prospective Technological Studies

Title: ICT and Regional Economic Dynamics: A Literature Review

Authors: Charlie Karlsson, Gunther Maier, Michaela Trippl, Iulia Siedschlag, Robert Owen and Gavin

Murphy.

Editors: Andrea de Pannizza and Marc Bogdanowicz Luxembourg: Publications Office of the European Union

2010

EUR - Scientific and Technical Research series - ISSN 1018-5593

ISBN 978-92-79-16568-9 doi:10.2791/46419

Abstract

The objective of this report is to review the relevant theoretical and empirical literature to provide a conceptual and methodological background for the analysis of the consequences of ICT use and globalisation on the regional economies in the European Union. We highlight the key aspects of ICT as a general purpose technology, discuss the economic impacts of ICT diffusion from macro and micro perspectives, and examine the spatial consequences of ICT diffusion. We focus on regional innovation systems and globalisation in order to propose an organizing framework for the analysis of the impact of ICT diffusion on regional development.

How to obtain EU publications

Our priced publications are available from EU Bookshop (http://bookshop.europa.eu), where you can place an order with the sales agent of your choice.

The Publications Office has a worldwide network of sales agents. You can obtain their contact details by sending a fax to (352) 29 29-42758.

The mission of the Joint Research Centre is to provide customer-driven scientific and technical support for the conception, development, implementation and monitoring of European Union policies. As a service of the European Commission, the Joint Research Centre functions as a reference centre of science and technology for the Union. Close to the policy-making process, it serves the common interest of the Member States, while being independent of special interests, whether private or national.





